



Independent Development Evaluation African Development Bank

From experience to knowledge... From knowledge to action... From action to impact

Impact Evaluation of the AfDB-funded Ghana Fufulso-Sawla Road Project

Summary Report

AFRICAN DEVELOPMENT BANK GROUP

THE CAME DE DE DER OF

August 2020

IDEV conducts different types of evaluations to achieve its strategic objectives







Independent Development Evaluation African Development Bank

From experience to knowledge... From knowledge to action... From action to impact

Impact Evaluation of the AfDB-funded Ghana Fufulso-Sawla Road Project

Summary Report

AFRICAN DEVELOPMENT BANK GROUP

THE CAME DE DE DER OF

August 2020

ACKNOWLEDGM	ENTS
Task Managers:	Joseph Mouanda, Chief Evaluation Officer, Task Manager, IDEV.1. Eric Kere, Senior Evaluation Officer, Co-Task Manager, IDEV.2.
Team member:	Clement Mensah, Consultant, IDEV.1.
Consultants:	Catholic University of America: Maria Sophia Aguirre; Hirut Hedamu; Kathryn Kroeker; and James Warburton Ablin Consult: Paulina Agyekum; Evans Tutu Akosah; Wilfred Sam-Awortwi; John Kwabo Boateng Kwaita as well as their team of enumerators
Internal peer reviewers:	Andrew Ajuang Anguko, Chief Quality and Methods Advisor, IDEV.0; Amadou Boly, Special Assistant, Office of the Chief Economist & Vice President and Girma Kumbi, Chief Evaluation Officer, IDEV.2.
External peer reviewers:	International Initiative for Impact Evaluation (3ie)
Internal Bank Reference Group:	COGH, PICU, Office of the Chief Economist & Vice President
Knowledge management officers:	Jayne Musumba, Principal Knowledge Management Officer, IDEV.3; Dieter Gijsbrechts, Senior Knowledge Management, Communications, and Events Officer, IDEV.3; and Tomas Zak, Junior Consultant, IDEV.3.
Other assistance provided by:	Myrtha Diop, Senior Administrative Assistant, IDEV.0; Henda Ayari, Team Assistant, IDEV.1; and Ruby Adzobu-Agyare, Secretary, IDEV.1.
Special thanks to:	Marie-Laure Akin-Olugbade, Director General, RDGW; Yero Baldeh, Director, RDTS; Jean Kizito Kabanguka, Division Manager, PICU1; Lydie Ehouman, Transport Economist, RDGW; Mam Tut Wadda, Chief Investment Officer Private Sector, RDGE; Akyea Sheila Enyonam, Senior Transport Infrastructure Engineer, COGH; Bumi Camara, Senior Country Economist, COGH and the entire team of COGH • Members of households, educators, health staff, and tourism professionals, who served as samples for this study • All stakeholders met at national and local levels for sharing constructive and insightful thoughts on the project and the sector • Members of the Ghana Statistical Services, especially Samuel K. Annim, Government Statistician and Abena Osei-Akoto, Director, Census and Survey
Division managers:	Rufael Fassil Foday Turay (OIC prior to 12/2019)
Evaluator General:	Roland Michelitsch

© 2020 African Development Bank Group All rights reserved – Published December 2020

Impact Evaluation of the AfDB-funded Ghana Fufulso-Sawla Road Project - Summary Report

IDEV Impact Evaluation, August 2020

Disclaimer

Unless expressly stated otherwise, the findings, interpretations and conclusions expressed in this publication are those of the various authors of the publication and are not necessarily those of the Management of the African Development Bank (the "Bank") and the African Development Fund (the "Fund"), Boards of Directors, Boards of Governors or the countries they represent.

Use of this publication is at the reader's sole risk. The content of this publication is provided without warranty of any kind, either express or implied, including without limitation warranties of merchantability, fitness for a particular purpose, and non-infringement of third-party rights. The Bank specifically does not make any warranties or representations as to the accuracy, completeness, reliability or current validity of any information contained in the publication. Under no circumstances including, but not limited to, negligence, shall the Bank be liable for any loss, damage, liability or expense incurred or suffered which is claimed to result directly or indirectly from use of this publication or reliance on its content.

This publication may contain advice, opinions, and statements of various information and content providers. The Bank does not represent or endorse the accuracy, completeness, reliability or current validity of any advice, opinion, statement or other information provided by any information or content provider or other person or entity. Reliance upon any such opinion, advice, statement, or other information shall also be at the reader's own risk.

About the AfDB

The overarching objective of the African Development Bank Group is to spur sustainable economic development and social progress in its regional member countries (RMCs), thus contributing to poverty reduction. The Bank Group achieves this objective by mobilizing and allocating resources for investment in RMCs and providing policy advice and technical assistance to support development efforts.

About Independent Development Evaluation (IDEV)

The mission of Independent Development Evaluation at the AfDB is to enhance the development effectiveness of the institution in its regional member countries through independent and instrumental evaluations and partnerships for sharing knowledge.

Independent Development Evaluation (IDEV)

African Development Bank Group Avenue Joseph Anoma, 01 BP 1387, Abidjan, 01 Côte d'Ivoire Phone: +225 20 26 28 41 E-mail: idevhelpdesk@afdb.org idev.afdb.org

Design & layout: CRÉON – www.creondesign.net Photo credit: AfDB Projects on Flickr

Contents

Abbreviations and Acronyms Executive Summary Management Response	v 1 9
Introduction	15
The Fufulso-Sawla Road Project	17
Methodology	23
Major Findings/Impacts of the Project	27
Impacts on transport-related outcomes Impact on access to socio-economic services Socio-economic impacts of the project Impacts on poverty reduction Impacts on institutional networks and relationships Unintended impacts of project interventions Sustainability of the project benefits	27 28 34 37 38 39 40
Conclusions, Lessons, and Recommendations	45
Conclusions Lessons Recommendations	45 47 47
Annexes	51

Contents

List of figures

Figure 1	Geographical distribution of ancillary interventions along the road	18
Figure 2	Project time-line relevant external events	19
Figure 3	Tourism normal distributions 2012–19, treatment vs control	32

List of tables

Table 1	Key design features of the FSRP	20
Table 2	Daily traffic flow for top four transport modes Fufulso-Larabanga-Grupe	27
Table 3	Summary effects and impact of the FSRP – transportation outcomes	28
Table 4	Summary effects and impact of the FSRP – market conditions outcomes	31
Table 5	Summary effects and impact of the FSRP – socio-economic services	34
Table 6	Summary effects and impact of the FSRP – household incomes	34
Table 7	Summary effects and impact of the FSRP – socio-economic outcomes	35
Table 8	Summary effects and impact of the FSRP – poverty reduction outcomes	38

Abbreviations and Acronyms

AfDB	African Development Bank	KVIP	Kumasi Ventilated Improved Pits
ANCOVA	Analysis of Covariance	MPI	Multidimensional Poverty Index
CDC	Centers for Disease Control and	OIC	Officer in Charge
0001	Chana Caustry Office	PA	Project Area
	Ginaria Country Onice	PICU	Infrastructure, Cities and Urban
FSRP	Fuluiso-Sawia Road Project		
GLSS	Ghana Living Standards Survey	RCI	Randomized Control Irial
GSS	Ghana Statistical Services	RDGE	Regional Directorate General for East Africa
GIS	Geographic Information System	RDGW	Regional Directorate General for
IDEV	Independent Development Evaluation		West Africa
KM	Kilometer	RDTS	Transition States Coordination Office



Executive Summary

Introduction

The African Development Bank (AfDB or "the Bank") prioritizes transport infrastructure as a critical means of achieving sustainable economic growth and reducing poverty. Transport forms a key focus area in the Bank's Ten-Year Strategy (2013-2022) core operational priorities. The road sub-sector makes up the bulk of the Bank's transport sector investments. funding the construction of national roads and major international corridors. The Ghana Fufulso-Sawla Road Project (FSRP) is one such investment. The overall thrust of the Fufulso-Sawla Road Project was to support the improvement of the investment environment through transport infrastructure development comprising the construction of a 147.5 km road as well as the provision of ancillary works along the main road corridor including: (i) rehabilitation and expansion of the main District Hospital at Damongo; (ii) construction/rehabilitation of eight main health centres; (iii) education facilities; (iv) construction of Moanori Bridge to better provide all-weather access for transportation of agricultural inputs and produce: (v) construction of a water treatment plant and borehole and; (vi) construction of four main market centres. The road project is located along a transit corridor linking landlocked countries (Burkina Faso, Mali, and Niger) in the north to the coastal Tema Port in Ghana, and providing access for improved trade between Ghana and its northern neighbors. To address key developmental challenges faced in the Project Area (PA) in the northern part of Ghana, additional interventions aside the main road corridor work included access roads, hospitals, schools, markets, water supply, etc., to generate positive benefits to all aspects of human development. The project, worth UA 110.58 million, was approved in 2010 and completed in 2015. It aimed at enhancing accessibility along the Fufulso-Sawla Road and improving livelihoods in the PA of influence. The project stands out as a flagship one in terms of its inclusive and integrated design for providing a holistic response to the socio-economic needs of the beneficiary districts. This summary report presents the findings, conclusions, lessons, and recommendations of the impact evaluation of the AfDB's support for a road transport intervention in Ghana-the Fufulso-Sawla Road Project. The summary report is prepared based on detailed technical reports.

What was evaluated

The Independent Development Evaluation (IDEV) conducted an impact evaluation of an integrated infrastructure project—the Fufulso-Sawla Road Project in Ghana. The evaluation estimates the average impact of all components of the project on development outcomes. The impact estimation results reflect the combined impact of all the components of the project.

Purpose of the evaluation

The purpose of this impact evaluation is to generate lessons and provide recommendations to maximize the impacts of ongoing and future inclusive and integrated infrastructure projects. The specific objectives are: (i) to estimate the impacts of AfDB supported integrated infrastructure proiects on key intermediate and long-term outcomes; (ii) to identify explanatory factors that affect the development outcomes of the project; and (iii) to generate lessons and provide recommendations for improving the impacts of ongoing and future integrated infrastructure interventions. The overarching evaluation guestion is: "What are the differences made by the Bank-supported integrated infrastructure project in Ghana?"

Methodology used

This impact evaluation applied a mixed-methods approach, and quantitative and qualitative approaches, in addressing the evaluation questions. In the quantitative studies, an attempt has been made to estimate the impacts of the road project on key variables of interest (traffic intensity; travel time and transport cost; and access to socioeconomic services, such as education, health, water supply, markets, and touristic attractions; and long-term outcomes, such as poverty, health, education, and employment) using household and community surveys. The qualitative study helped to provide further insights into contextual issues and perspectives with strong relevance for the design of the impact evaluation that enable or constrain the effectiveness and sustainability integrated road project interventions. of Quantitative methods included household surveys in 2.393 households across 17 treatment and 13 control communities. Furthermore, the secondary data collection involved an analysis of six household-level national surveys obtained from the Ghana Statistical Services (GSS) at the enumeration levels, and tourism data from Mole National Park authorities.

To establish evidence of the accurate project impact, it is necessary to compare an outcome on project beneficiaries with the counterfactual-a hypothetical outcome that would have been achieved in the absence of the intervention. Thus, the central issue in impact evaluations is to appropriately estimate the counterfactual, which cannot be directly observed. While the random assignment of the intervention is an ideal way to estimate a credible counterfactual, it is not always feasible to randomly assign the intervention, particularly for infrastructure projects, and this project is no exception. In the absence of an appropriate baseline, the evaluation used quasiexperimental methods of evaluation to measure the impacts of road project interventions. The evaluation employed the following strategies to have an appropriate estimate of the counterfactual.

We carefully selected a comparison road that is as "similar" to the target road as possible, based on observable characteristics, but that was not improved by the project. For the control group, the road between Yendi and Tatale was selected. This main road has similar characteristics to that of the Fufulso-Sawla road conditions before the FSRP implementation. It has urban areas, towns, and rural villages along the road, comparable to the treatment group, as well as proximity to a neighboring country. Communities included in the control and treatment groups were selected randomly from treatment and control district groups within four identified strata (urban areas, towns, villages along the road, and villages removed from the road). Balance tests were carried out to ensure comparability between the treatment and control groups. Overall, estimations statistical comparability suggest between communities selected along the Yendi-Tatale Road and those selected along the Fufulso-Sawla Road, as they show similar characteristics before the FSRP implementation.

In order to carry out the analysis, the evaluation utilized Ordinary Least Squares (OLS), analysis of covariance (ANCOVA), regression discontinuity, and interrupted time series analysis estimations in the context of an integral economic development approach. Underpinning this approach is the recognition of an individual's social dimension, which therefore considers the interpersonalrelational dimension of economic actions, i.e., the ways in which people interact to help or jeopardize sustainable development. The holistic design of the evaluation allows for the assessment of both the direct and long-term impacts of the intervention on households and communities. Furthermore, balanced test outcomes confirmed the comparability of the control and treatment groups.

Evaluation limitations and mitigation strategies

As with any evaluation, this evaluation inevitably has some limitations. This evaluation focused on "micro" impacts that were brought about by the project to households and/or individuals along the road project. Although it is important to examine potential "macro" impacts of the improvement of the road, such as enhancement of overall regional economic activities beyond the areas around the road project, it is beyond the scope of this evaluation. Second, the results of the estimation have some potential limitations highlighted by the literature and identified during the scoping mission. These limitations were taken into consideration to ensure unbiased estimates and adequate attribution of outcomes. These included: (i) the absence of appropriate baseline data: (ii) time inconsistency bias controlled through the use of control variables and panel regressions estimated with state-time fixed effects for time and communities; (iii) self-selection bias, addressed by selecting a counterfactual group with similar characteristics as the group treated; and finally (iv) results show only the average effect of all the components of the project, while the income declaration in the household survey has its limitations, which required caution in interpreting income outcomes. The inclusion of other aspects, such as a multidimensional poverty index (MPI), allows for a triangulation of the results.

Findings

Based on the theory of change, in this evaluation, we examined the key variables of interest. The estimated impact of the project on these outcomes are the following:

What was the net effect of the road on the intensity of the traffic, travel time, and travel cost?

The evaluation finds positive results of the impact of the FSRP on improved transportation conditions in beneficiary communities. It estimates that the FSRP led to a positive and statistically significant reduction in commuting time (by 33 percent) in 2019, equivalent to a reduction in commuting time of 120 minutes per month. The construction of the road significantly

reduced households' commuting time for accessing key facilities such as markets, clinics, hospitals, and schools.

The evaluation also finds that traffic flows within major towns of project beneficiary districts increased significantly. The evaluation estimates that average daily traffic flow across three main stations – Fufulso, Larabanga, and Grupe – increased by 121 percent in 2015. However, findings show that the project led to a significant increase in transport costs by 14.5 percent compared with the control group.

The road project improved domestic tourism markedly in the beneficiary districts. The findings show that average households reported improvements in tourism activities within beneficiary districts by 25 percent. Evidence from the discussions with key informants revealed that the road project, including the construction of an access road to the national park, significantly improved accessibility, even during rainy seasons.

What were the net effects of the road and related ancillary works on household income, employment, and access to social and economic services?

The quantitative analysis revealed that the road and related ancillary works indeed had positive impacts on a range of outcomes and affected livelihoods in different ways.

The road project interventions improved market conditions in beneficiary communities. Market conditions improved, including households' access to both farming and non-farming opportunities, inputs of production, and access to agricultural and other skills training, with an average increase of 14 percent in 2019, as a result of the road project.

The evaluation finds positive results of the impacts of the interventions on market integration and diversification. The FSRP led to a

significant increase in the market integration index (by 7 percent) and market diversification (by 2.2 percent) in 2019. Specifically, households that received assistance with business development achieved better integration into other markets than those that did not. As was the case for previous economic outcomes analyzed, findings on the market's diversification lend support for comprehensive interventions rather than singled-focused ones.

The evaluation finds positive results on the development/arrival of new business. The project led to a significant increase in the development and arrival of new business (by 12 percent). However, the impact on employment opportunities for both adults and youth was only significant when the road was associated with health facilities.

The road project had a significant positive impact on household incomes. The findings show that the project had significant positive effects on household incomes in 2019 (a monthly increase of around US\$68 or 84 percent from the mean in 2012). The impact was not uniform across ancillary interventions. Evidence of a more positive impact on household incomes was found in communities where the road construction was accompanied by health-care facilities (an increase of US\$18 or 100 percent). As expected, communities closer to the road reported higher increases of income, but the difference was not statistically significant.

Access to quality water supply and sanitation improved due to the road project. The project significantly improved households' access to a quality water supply. The water quality index among beneficiary households increased by 17 percent in 2015, compared with the control group. This improvement is partly accounted for by the water treatment plant that accompanied the road project, which covered some 10,000 people. Also, the evaluation found a statistically significant impact of the project on households' sanitation conditions, yielding an average improvement in sanitation conditions index among beneficiary households of 14 percent in 2019. The integrated transport project had a positive impact on health and education outcomes. The findings show statistically significant positive effects of the transport project on the overall health of beneficiaries, increasing by 9 percent in 2015 compared with control households. No statistical evidence of the construction of the road was found on child mortality or prenatal care, apart from households alongside the road, for which a significant increase of 41 percent was observed in terms of prenatal care. The evaluation found statistically significant positive effects on education outcomes. Children who benefited from the new road when they were in primary school completed 5.5 grades more than those who did not. If they were living in urban areas, they completed 11 years more education than those who lived in rural areas. Also, children who were of schoolage when the road was constructed had a 3.96 times higher probability of attending school.

The evaluation demonstrated that the Banksupported integrated road project had the desired effects on poverty reduction. The multidimensional poverty definition used in the study considers that poverty reduction can come from changes in sources of income and better access to social services. Statistical evidence showed that with the road only, the MPI decreased by 2.16 percent among beneficiary households compared with control households in 2015, and by 2.59 percent in 2019. These results show that the project had significant positive effects not only on the living standards of beneficiaries but also on education and health outcomes. In communities where road construction was accompanied by schools and market centers, impacts on household poverty reduction were amplified. This underscores the significance of adopting integrative approaches in road projects, especially those targeting poor and economically disadvantaged areas.

How were the costs and benefits distributed?

The ancillary works generated a proportionately greater additional effect than their additional costs (8.2 percent of the total project cost). As

indicated, road construction alone was estimated to have led to a statistically significant reduction in the MPI among beneficiary households by 2.16 percent in 2015. Households that benefited from a school in addition to the road experienced a significant additional reduction in the MPI by another 0.54 percent (to 2.7 percent), or 20 percent of the total effect. The construction of a market in addition to the road led to an additional statistically significant reduction in MPI by 0.39 percent (to 2.55 percent), or 15 percent of the total effect. The ancillary works thus generated a proportionately greater additional effect on the MPI than their additional cost.

The intervention benefited women and girls specifically, albeit to a lesser degree than men. Indeed, there is evidence of males using the road more than females, but females with higher levels of education also used the main road more frequently by 2019. The project also led to an additional reduction by 0.25 percent in the MPI for male- versus female-headed households. Males also benefited more than females from the positive impact on education effects of the road. However, female school enrolment and retention increased, maternal death decreased, access by women to local markets and water collection improved, and their economic opportunities also expanded.

What were the unintended impacts?

The project promoted greater social cohesion among beneficiary communities. The equitable distribution of the ancillary facilities brought greater social cohesion and peace to communities along the road corridor. Indeed, inter-community conflict decreased because communities depend on each other for varying social services. On security, the project contributed to reduced highway robbery.

The project also helped to catalyze investments from other development partners to build on its outputs and outcomes. The World Bank is expanding the network of the water treatment plant—one of the many ancillary interventions supported under the FSRP—with an amount of US\$1 million. On unintended negative consequences, qualitative interviews point to an impact on environmental degradation. The project negatively affected the environment due to marked growth in charcoal-burning activities in the three beneficiary districts, although they also positively impacted district assemblies through the generation of higher revenues from the booming charcoal-burning business. Also, the construction of the road facilitated Rosewood logging within the project's enclave - a phenomenon that continues to threaten the Mole National Park, in particular.

Are development benefits from the projects sustainable?

The sustainability of development outcomes of the Bank-supported road project and related ancillary works were highly unlikely. Regular maintenance is a critical precondition for sustaining the positive impacts that road and ancillary facilities bring to communities. Under this project, however, the field visit and interviews with beneficiaries revealed that poor maintenance affected many of the facilities provided, especially health centers, schools, and bungalows built for teachers and nurses. For instance, solar panels installed to power some health centers were either burgled or were not fully functional. For the water treatment plant, no maintenance plan was in place because of the under-utilization of the plant. This problem was further aggravated by a lack of maintenance plans at the local government authority level. For the road, the Larabanga-Sawla section was degraded even before its delivery. reducing the impact of the project. Several reasons contributed to this: poor scheme design. weak organizational and institutional capacity, and the lack of active community and government ownership of the project.

Lessons

The following are the key lessons from this impact evaluation.

6

Lesson 1: Integrating community development interventions into road transport infrastructure brings an added value to, and fast tracks, the Bank's poverty reduction, job creation, and inclusive development efforts.

Results from the evaluation showed that combining road construction with the construction of market centers and schools in a largely rural context yielded additional positive impacts on multidimensional poverty reduction. This underscores the need to pay close attention to the context when designing such integrated road interventions.

Lesson 2: Relational factors are key to the sustainability of integrated road interventions such as the FSRP, as they can amplify or jeopardize a project's impact.

For example, the failure to ensure active participation of project beneficiaries beyond 'information sharing' undermined the maintenance of the ancillary facilities and thus affected sustainability. Conversely, greater multisector collaboration demonstrated among staff of the Bank's Ghana country offices contributed to the success of the project by drawing on the availability of the right caliber and mix of staff. This lends credence to the important role of the Bank's Development and Business Delivery Model (DBDM) which, among others, underscores the importance of having the full complement of staff to support its operations in Regional Member Countries (RMCs).

Lesson 3: Coupling road projects with community development interventions can increase the likelihood of benefits accruing to women and girls.

In particular, the evaluation demonstrated marked gains regarding women's and girls' access to social services such as health and education. This form of integrated intervention can add a layer of gender sensitivity to road interventions.

Recommendations

IDEV makes the following recommendations:

Recommendation 1: Enhance the Bank's integrated approach to its road investments to foster development impact in terms of poverty reduction.

Evaluation results showed that a road alone is not enough to tackle poverty. The poor, lacking assets to take advantage of better opportunities that a road may bring, benefit from additional support. This suggests that integrated projects are necessary to tackle poverty effectively. Indeed, while the inclusion of community development projects in road projects such as the FSRP make up a small fraction of overall cost, they can make a significant difference, particularly for increasing access to social services. Such projects provide an impetus for amplifying the social impacts of roads, which in the long run brings an added value to multidimensional poverty reduction and inclusive development. Thus, first, the Bank should explore more of these integrated road projects and proactively adopt them as flagships for its inclusive growth, poverty reduction, and rural development efforts. Second, it should, on the back of its One-Bank approach, step up support for community development components of road projects by deliberately committing more financial resources to them and not treating them as add-ons or afterthoughts. However, the Bank should also be cognizant of the fact that such approaches may not work in all contexts and, hence, should adapt the choice of the ancillary works/services to local realities.

Recommendation 2: Improve the quality of road projects' design and results focus.

Given the multidimensional nature of such interventions, having a theory of change that demonstrates causal relationships and impact linkages is critical. This will facilitate the design, implementation, and evaluation of the interventions. The Ghana FSRP used an integrated approach. However, the project documents did not clearly present how the road will interact with the ancillary works to achieve the expected intermediate outcomes. Therefore, for similar future investments: (i) the Bank should base its integrated road intervention design on an evidence-based well-articulated theory of change, with a clear pathway through which the logic model would occur; and (ii) the baseline for the treatment and control groups should be established during the design phase of the intervention for such large-scale, innovative, and flagship projects to determine not only whether an intervention is effective, but also to compare options for making interventions more effective.

Recommendation 3: Strengthen the human and institutional capacity to sustain development gains.

Neglect of periodic maintenance threatens longterm social and economic benefits from the road and ancillary facilities. The Bank should strengthen the human and institutional capacity for the sustainability of road projects and their ancillary facilities by: (i) engaging in dialogue with the government to explore partnershipbased approaches where a memorandum of understanding can be signed with local government authorities on maintaining the facilities; (ii) encouraging meaningful beneficiary participation and contribution in all infrastructure maintenance activities, including labor and material inputs, thus creating ownership through risksharing, to guarantee the sustainability of actions; and (iii) adequately mitigating or minimizing unintended environmental degradation.



Management Response

Management welcomes the independent evaluation of the Fufulso-Sawla Road Project (FSRP) in Ghana, four years following the project's completion. The FSRP can be described as an integrated project that addressed transport, education, health, agriculture, water and sanitation, trade, and tourism. The report makes good recommendations and assessments that will enhance the Bank's ongoing efforts to render its operations more inclusive and address development challenges more holistically, particularly in rural areas. Overall, Management agrees with the evaluation's findings and recommendations, especially the need to deepen projects' integrated approach, improve the quality of project design at the preparation stage, and develop institutional and human capacity to sustain the Bank's investments. The independent evaluation generates knowledge that will improve the design and impacts of similar interventions in future.

Introduction

Over the years, Ghana's transport sector has been guided by government policies and strategies designed to provide efficient transportation services in order to catalyse economic growth. The sector draws on the National Transport Policy, the Integrated Transport Plan, and national medium-term plans rolled out by the government every four years following the election cycle. The country's medium-term development agenda seeks to make Ghana the transportation and logistics hub of West Africa, providing air, road, rail, and sea linkages to countries in the subregion, especially Ghana's landlocked neighbours in the Sahel.

For its part, the Bank prioritises transport infrastructure as a critical means of achieving sustainable economic growth and reducing poverty. Transport is a key focus of the Integrate Africa priority and the Bank's Ten-Year Strategy 2013–2022.

The Fufulso-Sawla Road Project (FSRP) was aligned with the Bank's and Ghana's strategies in this regard. It aimed to improve transport supply with a view to promoting the integration of the neighbouring regions within the national economic space and boosting inter-regional trade and tourism.

To address key developmental challenges in northern Ghana, the Bank collaborated with the Government of Ghana to pilot an integrated approach that added interventions to the main road project. These interventions consisted of rural feeder roads, health centres, schools, markets, and water and sanitation facilities, all of which benefited populations in the region. The road project is located along a transit corridor that links landlocked countries in the north (Burkina Faso, Mali, and Niger) to Tema Port on Ghana's coast. The project made it possible to improve trade between Ghana and its northern neighbours.

Management welcomes the independent evaluation. Not only will it inform future country strategy papers (CSPs) for Ghana, it will also shape the design of future road projects, helping the Bank meet its objective of preparing holistic transport projects as a means of improving impacts in the sector. Management takes note of the following lessons:

- Integrating community development interventions into road transport infrastructure adds value and fast-tracks poverty reduction, job creation, and inclusive development. These elements are key to sustaining integrated road interventions such as the FSRP. How a project treats these elements can amplify or jeopardise the project's impact.
- Adopting an integrated approach to a project like this one, means including sectors beyond transport. To manage this, projects should develop

a framework that project teams can use to capture all sector inputs at the beginning, monitor them as the project proceeds, and evaluate them at the end. Such a framework would make it possible for teams that cut across several sectors to work more effectively together. This is also in line with the Bank's selectivity approach to prioritise fewer and larger projects.

- Coupling road projects with community development interventions can increase the likelihood of benefits accruing to women and girls. The independent evaluation documented marked gains in women's and girls' access to social services such as health and education. This demonstrates that integrated interventions can add a layer of gender sensitivity to road projects.
- Project sustainability is still very much an issue. Management must do more to make sure that when projects are being prepared, the investments—whether in transport or in other sectors—will be secured and protected after the project has been delivered.

Main findings

Transport-related outcomes

Management welcomes the evaluation's findings with regards to two of the intervention's impacts: the increase in traffic and the reduction of travel times and travel costs. The improvement in road conditions developed several economic activities and increased traffic between the towns and villages along the project corridor and within the project's zone of influence. Notable, too, is the 33% reduction in travel times. In addition, upgrading the road boosted inter-regional/ inter-district trade and facilitated exchanges between areas of agricultural production and market centres. Greater access to inputs also caused agricultural productivity to grow.

In addition, upgrading the road significantly reduced households' commuting time to essential facilities

such as market centres, clinics, hospitals, and schools. Before the road project, only big buses could travel. After road conditions improved, private cars, pickups, and taxis could also use the road, carrying commuters between towns or bringing tourists to visit destinations along the road corridor. The independent evaluation also flagged an increase in travel fares, but in fact, fares decreased in real prices. More specifically, inflation averaged 12.9% per year between 2014 and after the project was completed in 2019. The fare in 2014 was US \$3.5, equivalent to US \$6.40 in real prices, compared to US \$5.25, equivalent to US \$5.70, in 2019. The value of the fare in 2014 as of 2019 is US \$5.90 and still shows a reduction as compared to the real fares of US \$5.25.

Management is of the view that the independent evaluation should have considered that inflation and competition between transport operators played a key role in prices.

Socioeconomic impacts

The project's impact on households' socioeconomic conditions is worth mentioning. Upgrading the road reduced poverty amongst beneficiary households. It also improved employment opportunities for adults and youth at a statistically significant level. In addition, local communities benefited from better socioeconomic infrastructures such as schools, markets, and healthcare facilities, which produced an additional positive and significant impact on occupational opportunities.

Management takes note of the improvement in market conditions—including households' access to farming and non-farming opportunities, inputs of production, and agricultural areas—which continued four years after the project was completed.

The patronage of tourism amongst Ghanaians also experienced a marked increase in the beneficiary districts. Households reported more tourism activity and interviews of key tourism stakeholders revealed new developments and investments in the hospitality industry as a result. Management welcomes the evaluation's confirmation of the findings disclosed in the project completion report.

Management has always held the view that transport infrastructure enables socioeconomic change: that once a road opens to a remote opportunities for community. manv more interventions in other sectors ensue. Households experienced a general improvement in their health and access to social amenities because the project did not just provide a road, but also supplied other socioeconomic infrastructure facilities that are critical to improving the quality of life. Management notes that the FSRP significantly improved women and girls' day-to-day activities. Although men may have benefited more than women, female school enrolment and retention nonetheless increased. maternal death fell significantly, access to local markets and water collection improved, and economic opportunities expanded for women and girls. These impacts underscore the importance of the transport sector developing projects using an integrated approach.

Impacts on institutional networks and relationships

Management noted that the project's ancillary works generated an effect on the Multidimensional Poverty Index (MPI) that was proportionally greater than the expenses they cost the project. However, Management also acknowledges the evaluation's finding that the project's integrated approach failed to ensure that social amenities were properly maintained. In practice, the sustainment of ancillary facilities depends on such factors as inclusion of the facilities in sectors' maintenance programmes, on ownership, and on training staff to operate the facilities or equipment. Much effort was made to engage communities in the design and implementation of the ancillary works, and the Bank ensured that stakeholders were engaged adequately during the preparation and appraisal of the project. Yet no formal structure was put in place to sustain the commitment to maintain the investments. This is an important lesson.

Management acknowledges the need for a framework to guide the formulation and design of integrated infrastructure projects that cut across sectors to resolve the issues highlighted in the independent evaluation and generate greater impacts and sustainability.

Unintended impacts

Strong catalytic leverage and greater social cohesion amongst beneficiary communities are two of the positive unintended results generated by the road project. The project's integrated approach helped catalyse investments from other development partners to build on the project's outputs and outcomes. More specifically, the World Bank agreed with the Government of Ghana to expand Ghana's network of water treatment plants, one of the many ancillary interventions supported under the FSRP.

During project design and formulation, the Government of Ghana and the Bank made every effort for the project to be as inclusive as possible. Every town and settlement along the road corridor benefited not only from the road but also from other socioeconomic infrastructure such as schools, health facilities, and water services. Some zones outside the project area also benefited from the water treatment plant, because of the plant's location. The plant has the capacity to accommodate future expansions, which have already begun.

Management notes that although the resources allocated to ancillary works were small, their benefits have been considerable.

Conclusion

Management agrees with the findings of the independent evaluation. The lessons learnt from the FSRP will improve the formulation, design, and implementation of future transport projects. The Management Action Record, below, sets out specific actions against the recommendations made in the evaluation report.

determine not only whether an intervention is effective, but also to compare options for making interventions more effective.

Management Action Record						
Recommendation	Management Response					
Recommendation 1: Enhance the Bank's intimpact in terms of poverty reduction.	egrated approach to its road investment to foster development					
 a. The Bank should explore more of these integrated road projects and proactively adopt them as flagships for its inclusive growth, poverty reduction, and rural development efforts. b. The Bank should, on the back of its One-Bank approach, step up support for community development components of road projects by deliberately committing more financial resources to them and not treat them as add-ons or afterthoughts. Besides, the Bank should also be cognizant of the fact that such approaches may not work in all contexts and, hence, should adapt the choice of the ancillary works/ services to local realities. 	Agreed. Management broadly agrees with the recommendation; however, governments need to lead the process and include the preparation of integrated projects in their development agenda. On the Bank side, Management is of the view that integrated designs should consider including infrastructure that will support the developmental logistics chains (e.g., bus stations, storage facilities, markets centres, and sanitary facilities). Transport projects should also be measured in terms of both their transport-related outcomes and their socioeconomic impacts on people's quality of life. However, this recommendation may not apply everywhere; for example, it may not apply to middle-income countries. Further Action: 1. Management will develop an operational framework to guide integrated road infrastructure projects. This Framework will be tailored to suit each country based on their specific dynamics. The framework will describe how to design and implement projects across sectors using multi-disciplinary project teams and how the whole programme will be measured for success. Developing the framework will involve collaboration between the infrastructure sector and such sectors as water and sanitation, agriculture, education, and trade. (PICU, Q4 2022)					
Recommendation 2: Improve the quality of	road projects design and results focus.					
a. The Bank should base its integrated road intervention design on an evidence-based well-articulated theory of change, with a clear pathway through which the logic model would occur; and	Agreed . Management agrees with the recommendation. Countries will be advised and supported from the onset to prepare integrated infrastructure projects that can easily be appraised. Management will also consider revising the theory of change for the road sub-sector and the results measurement framework to better account for integrated projects. This will					
b. The baseline for the engagement of control groups should be established during the design of a project phase of the intervention for such large scale, innovative, and flagship projects to	 improve the quality and readiness of projects before appraisal. Further Action: 2. Revise the Bank's standard terms of reference for road studies to provide for an integrated approach, including the 					

collection of baseline data. (PICU Q2 2021)

	Management Action Record					
	Recommendation	Management Response				
	Recommendation 3: Strengthen the huma	n and institutional capacity to sustain development gains.				
;	 The Bank should engage in dialogue with the government to explore partnership- based approaches where a memorandum of understanding can be signed with local government authorities on maintaining the facilities; 	Agreed. Management broadly agrees with the recommendation. Although the maintenance culture has improved in most client countries, there is still a long way to go to preserve and sustain investments in infrastructure assets. One of the lessons learnt from this project is that stakeholders should be involved from the start in selecting sites and identifying socioeconomic infrastructure and priority activities by means of a more				
1	b. Encourage meaningful beneficiary participation and contribution in all infrastructure maintenance activities, including labor and material inputs, thus creating ownership through risk- sharing, to guarantee the sustainability of actions; and	participatory approach involving government representatives at the central and decentralised levels, community development committees, civil society associations, users, and production cooperatives. This will lead to agreement on how socioeconomic infrastructure will be managed and maintained. In addition, the Bank should assist countries to implement decentralisation reforms and empower local authorities/communities. Finally, the Bank will encourage greater involvement of the private				
(c. Bank ensures that the Borrower implements the disclosed Environmental and social documents in compliance to the Financing Agreement.	sector in the operation and maintenance of facilities through public-private partnerships. There will also be an agreement with decentralized authorities on how socioeconomic infrastructure will be managed and maintained.				
		 Further Actions: 3. Include in projects' logical framework a measure of stakeholder engagement on a project. (PICU Q4,2021). 				
		 Strengthen implementation support to borrowers/clients by providing technical assistance and check compliance during the implementation phase, as indicated in the approved Safeguards Strengthening Action Plan (SSAP). (PICU, Ongoing) 				



Introduction

The African Development Bank (AfDB or "the Bank") prioritizes transport infrastructure as a critical means of achieving sustainable economic growth and reducing poverty. Transport forms a key focus area in the Bank's Ten-Year Strategy (2013-2022) and core operational priorities. The road sub-sector makes up the bulk of the Bank's transport sector investments, funding the construction of national roads and major international corridors. The Ghana Fufulso-Sawla Road Project is one such investment. This summary report presents the findings, conclusions, lessons, and recommendations of the impact evaluation of AfDB's support of a road transport intervention in Ghana-the Fufulso-Sawla Road Project (FSRP). Several impact evaluations have been carried out on road construction projects in Sub-Saharan Africa and elsewhere in the world. These include road construction interventions by the AfDB. Studies of the impact of road construction on the multiple dimensions of poverty and economic development have been conducted, but are scarce.¹ This impact evaluation contributes towards filling this gap. This summary report is prepared based on detailed technical reports.

Purpose and objectives: The purpose of the impact evaluation is to generate lessons and provide recommendations to maximize the impacts of ongoing and future inclusive and integrated infrastructure projects. The specific objectives are to:

Estimate the impacts of AfDB supported integrated infrastructure projects on key intermediate outcomes (traffic intensity; travel time and transport cost; and access to socio-economic services, such as education, health, water supply, markets, and touristic attractions) and long-term outcomes (poverty, health, education, and employment);

- I Identify explanatory factors that affect the development outcomes of the project; and
- Generate lessons and provide recommendations for improving the impacts of ongoing and future integrated infrastructure interventions.

Evaluation questions: The overarching evaluation question is: "What are the differences made by the Bank supported integrated road project in Ghana?" Specific questions are:

- What is the impact (net effect) of the road on the intensity of the traffic, travel time, and travel cost?
- What is the impact of the road and related ancillary works on household income, employment, and access to social and economic services?
- What causal factors have resulted in the observed impacts?
- Has the intervention resulted in any unintended impacts?
- Are the impacts of the project differentiated by gender?
- What lessons can be learned from this intervention?

Scope: The primary focus was on the direct and intermediate outcomes realized among the final beneficiaries—the communities and households. However, this evaluation goes beyond the immediate and addresses changes in the behavior of the beneficiaries. In terms of budget, road construction represented 85.4 percent of the construction work carried out and ancillary works represented 7.4 percent. The remaining budget was spent

on studies (1.4%), project management (5.7%) and compensation and resettlement (0.05%). Proportionally, 36 percent of the ancillary works' budget was allocated to health facilities, 28 percent to education facilities, 2 percent to the grain storage built in Damongo, 9 percent to water supplies (boreholes), and 25 percent to markets and truck parks. The road construction began in 2012 and was completed in 2015. The impact data were collected in 2019. As per the project appraisal report (PAR), the primary and direct project beneficiaries of the project include the inhabitants near the road. The other project beneficiaries include diverse stakeholders

at local, regional, national, and international levels. The road corridor's immediate vicinity is inhabited by an estimated population of 30,000, of whom 50.8 percent are women.

This report presents the result of the impact evaluation. The rest of the report consists of the following sections: Section 2 provides a brief description of the project, which was covered by this impact evaluation. Section 3 describes the evaluation methodology. Section 4 presents the key findings of the evaluation. Finally, Section 5 summarizes the conclusions, lessons, and recommendations.

The Fufulso-Sawla Road Project

While road projects financed by the Bank have traditionally tended to be implemented as a stand-alone intervention, in recent years, this has changed, where in addition to the construction of roads, other complementary community development interventions are incorporated as part of the intervention package. The Fufulso-Sawla Road Project (FSRP) in Ghana is one such project, which aimed at implementing other socio-economic infrastructure (water, health, education, agriculture, and social protection) besides the construction of a 147.5 km road.

The overall thrust of the FSRP, which is examined in this impact evaluation, was to support the improvement of the investment environment through transport infrastructure development. The project directly supported the Ghana Shared Growth and Development Agenda (2010-2013). This Agenda sees transport infrastructure as one of the channels through which economic growth and poverty reduction is envisioned. It also considers transport as relevant in achieving the country's goal of becoming a leading agro-industrial country. The FSRP aims to address development challenges caused by poor road infrastructure in the former Northern Region of Ghana, now the Savannah Region, which is one of the poorest and most isolated regions of the country.² The FSRP links the West and East main highways of Ghana. The road construction extends across three districts: Central Gonja, West Gonja, and Sawla-Tuna Kariba, between Fufulso town and Sawla city. Together, these districts constitute a population of 272,467, out of which 78 percent reside in rural areas (Ghana Statistical Service, 2015). The road project is located along a transit corridor linking landlocked countries (Burkina Faso, Mali, and Niger) in the north to the Tema Port in Ghana and providing access for improved trade between Ghana and its northern neighbors. The road corridor's immediate vicinity is

inhabited by an estimated population of 30,000, of whom 50.8 percent are women. These constitute the direct beneficiaries of the project. Other beneficiaries are people from other parts of the Savannah Region, traders and road transport operators, and travelers from the Upper East and West Regions, as well as Burkina Faso, Mali, and Niger.

Despite their vast resources and potential, the three districts along the road corridor are among the most deprived areas in Ghana. The incidence of household poverty is high across the three districts: Central Gonja (61 percent), West Gonja (53 percent), and Sawla-Tuna-Kalba (63 percent) (Ghana Statistical Service, 2015), Only 30 percent of households along the road have access to safe water, far below the regional and the national averages of 58 and 57 percent, respectively. Most households rely on unsafe water sources, such as dug-out dams for drinking water, most of which dry up during the dry season. This has contributed to a high prevalence of water-borne diseases in the project area. Before the project, the three districts along the road had only one medical doctor based in Damongo Hospital. The districts had, on average, a doctor-patient ratio of 1:29,394, nearly threetimes the national ratio of 1:10,380. Educational infrastructure in many communities along the road is either inadequate or in a deplorable state of repair, with some classes held in sheds or under trees. Net primary school enrolment averages about 65 percent, below the national average of about 84 percent.

The AfDB specified six development objectives for this intervention: (i) improve the Fufulso–Sawla road condition; (ii) increase local economic activity resulting in improvement in livelihood; (iii) improve the mobility index in the Savannah Region and between Fufulso and Sawla in particular; (iv) reduce transport costs and travel times; (v) increase revenue and the number of tourists visiting the area; and (vi) increase access to socio-economic infrastructure (schools, health posts, markets, truck parks, and water and sanitation) for communities. The latter was expected to expand the road's effect by improving socio-economic factors.³ Specific synergies between the road and the ancillary works were not defined at the time.

At completion, the 147.5 km of road built between Sawla and Fufulso was divided into two sections, Lot 1 (Fufulso-Larabanga) and Lot 2 (Larabanga-Sawla). Each Lot also incorporated several ancillary works serving 15 communities along the entire corridor. The construction in Lot 1 was carried out by the China International Water & Electric Corporation. The company took charge and executed all road and ancillary works. The China Harbour Engineering Company Ltd was the construction contractor in Lot 2; the company subcontracted the ancillary works to a third party. The same consultant supervised the work in both lots. The constructors' contract included a 12 months defects liability period (DPL).

The road constructed was a Double Bituminous Surface Treatment (DBST), standard of a 7.3 m two-way one lane single carriageway plus 2.5 m shoulders on each side of the carriageway. The ancillary works built in the 15 communities located along the corridor varied across communities. They encompassed school facilities, health centers, and rehabilitation/renovation of the hospital in Damango, accommodations for nurses and teachers, Kumasi Ventilated Improved Pits (KVIPs), boreholes, truck parks, markets, and a grain storage facility.⁴ Proportionally, 36 percent of the ancillary work was allocated to health facilities. 28 percent to education facilities, 2 percent to the grain store built in Damango, 9 percent to water supplies (boreholes), and 25 percent to markets and truck parks.⁵

In addition to the Mognori Bridge, nine communities received educational and sanitation facilities (Janikua, Kojope, Busunu, Janokponto, Bonyanto,



Nasoyin, Jentilpe, and Nyange), seven received health and sanitation facilities (Sawla, Jentilpe, Nyange, Damango, Grupe, and Kabampe), and four communities received road safety and/or HIV/AIDS sensitization and awareness training (Grupe, Sawla, Larabanga, and Kabampe). The communities vary in terms of size and their degree of development, some being very small communities or villages, while others are small towns or small cities.

Figure 1 depicts the distribution of the different components along the constructed road. As might be expected, ancillary elements are concentrated where there is a higher population density, while markets are mostly evenly distributed along the larger towns/cities.

This intervention sought to improve existing poor road infrastructure in the region, thus opening new opportunities for local socio-economic development and trade with neighboring countries in the north. In December 2010, the Bank approved the financing of the Ghana FSRP. Although the project duration was between 2011 and 2015, the actual construction of the road and implementation of its accompanying ancillary facilities took place between January 2012 and July 2015, under the management of the Ghana Highway Authority (GHA), with collaboration from Ghana's Ministries of Health and Education in specific aspects relating to the ancillary works. The official commissioning of the road took place in August 2015.

Figure 2 depicts the timeline of the project and impact evaluation, as well as relevant external events. Among the latter is the election of President Mahama in December 2012, whose home place is located in the Savannah Region: the large Accra floods in June 2015, shortly before the completion of the construction of the Fufulso-Sawla road: and the election of President Akufo-Addo, the creation of the Savannah Region in December 2018, and the declaration of Damongo as its capital. The latter has accelerated the economic development of the FSRP area because of the growing presence of governmental administration offices and personnel. While presidential elections can be treated as an exogenous event that affected the country nationwide, and therefore both the treatment and control groups, this is not the case for the designation of Damongo as a capital for the newly created Savannah Region, However, Zabsugu, Tatale, and Yendi are among the communities included in the control group. All three towns are the three districts' capitals.

The main ancillary works built in the 15 communities located along the corridor varied



across communities and encompassed: (i) school facilities (9); (ii) health facilities (8) (Community Health-based Planning Services (CHPS) compounds and health posts); (iii) one accident center with an ambulance; (iv) markets and truck parks (4); and

(v) water supply (41 boreholes and one water treatment plant) and a grain storage facility.

The cost of the whole project was UA 110.59 million.⁶ Table 1 presents the key design features of the FSRP.

Table 1: Key design features of the FSRP

Domain	Est. Cost (UA million)	%	Component description
Road Construction Works	94.48	85.4%	Construction of 147.5 km of road between Fufulso and Sawla including lined drains at major settlements along the road.
			Environmental and social impact mitigation measures.
			Sensitization of project area inhabitants on HIV/AIDS and STIs; malaria and water-related diseases; road safety; and environmental protection.
			I Construction of infrastructure in support of tourism (parking spaces, signposts along the road and in the Mole National Park, construction of a conservation museum at the entrance of the Mole National Park).
			Construction of 6 km long access roads and driveways to socioeconomic infrastructures along the road.
			Construction of two permanent weighbridge stations.
			Works construction supervision.
Ancillary Works	8.21	7.42%	Water supply in the communities along the road corridor.
			Support for women's agro-processing activities along the road corridor.
			Rehabilitation of existing schools (classrooms, equipment/furniture, sanitation, boreholes, fencing) along the road corridor.
			Rehabilitation of existing health facilities: (i) Community Health-based Planning Services (CHPS) compounds (works, equipment, and solar panels); (ii) construction of an accident center midway of the road (at the existing district hospital at Damongo and the provision of ambulances).
			Construction of markets and truck parks in Larabanga, Busunu, Fufulso, Sawla, and Damongo.
			Rehabilitation of the grain storage area at Damongo.
			Design and supervision of ancillary works.
Studies	1.54	1.39%	Road studies.
Project Management	6.31	5.71%	Project coordination.
			I Technical assistance to the Project Coordination Unit.
			Project impact monitoring and evaluation.
			Financial and technical audit.
Compensation and Resettlement	0.05	0.05%	Compensation and resettlement costs for all project-affected persons
Total	110.59	100%	





Methodology

This impact evaluation applies a mixedmethods approach, and quantitative and qualitative approaches, in addressing the evaluation questions. In the quantitative studies. an attempt has been made to estimate the impacts of the road project on key variables of interest (traffic intensity; travel time and transport cost; access to socio-economic services, such as education, health. water supply, markets, and touristic attractions: and long-term outcomes, such as poverty, health. education, and employment) using household and community surveys. The qualitative study helped to provide more insights into contextual issues and perspectives with strong relevance for design of the impact evaluation that enable or constrain the effectiveness and sustainability of integrated road project interventions. In the case of quantitative methods. IDEV carried out household surveys in 2,393 households across 17 treatment and 13 control communities. Furthermore, the secondary data collection involved an analysis of six household-level national surveys obtained from the Ghana Statistical Services (GSS) at the enumeration levels and tourism data from Mole National Park authorities. The details of both quantitative and qualitative methods are presented in Annexes 2 and 3 of the Technical Annexes.

The evaluation design and estimation strategies:

In order to establish evidence of the accurate project impact, it is necessary to compare an outcome for project beneficiaries with the counterfactual—a hypothetical outcome that would have been achieved in the absence of the intervention. Thus, the central issue in impact evaluations is to appropriately estimate the counterfactual, which cannot be directly observed. While the random assignment of the intervention is an ideal way to estimate a credible counterfactual, it is not always feasible to randomly assign the intervention particularly for infrastructure projects, and this project is no exception. In the absence of appropriate baseline data, the evaluation used quasi-experimental methods of evaluation to measure the impacts of road project interventions. The evaluation employed the following strategies to have an appropriate estimate of the counterfactual.

The evaluation team carefully selected a comparison road that was as "similar" to the target road as possible based on observable characteristics, but that was not improved by the project. For the control group, the main road stretching between Yendi and Tatale was selected (Annex 6 of the Technical Annexes). This main road has similar characteristics to the Fufulso-Sawla road conditions before the FSRP implementation. It has the presence of urban areas. towns, and rural villages along the road, comparable to the treatment group, as well as proximity to a neighboring country. Communities included in the control and treatment groups were selected randomly from the treatment and control district groups within four identified strata (urban areas, towns, villages along the road, and villages removed from the road). Balance tests were carried out to ensure comparability between treatment and control groups. Overall, estimations suggest statistical comparability between communities selected along the Yendi-Tatale road and those selected along the Fufulso-Sawla road, as they show similar characteristics before the FSRP implementation.

In order to conduct the analysis, the evaluation utilized Ordinary Least Squares (OLS), analysis of covariance (ANCOVA), regression discontinuity, and interrupted time series analysis estimations in the context of an integral economic development approach. Underpinning this approach is the recognition of an individual's social dimension, which therefore considers the interpersonal-relational dimension of economic actions, i.e., the way people interact to help or jeopardize sustainable development. The holistic design of the evaluation allows for the assessment of both the direct and long-term impacts of the intervention on households and their communities. Furthermore, balance test outcomes confirmed the comparability of the control and treatment groups. However, although we carefully selected similar comparison roads, this does not ensure that the comparison group is similar to the treatment group in terms of unobservable characteristics as well. because the similarity was determined by a few observed variables. Thus, as a second strategy, the evaluation utilized OLS, ANCOVA, regression discontinuity, and interrupted time series analysis estimations in the context of an integral economic development approach for the impact analysis of the road's and ancillaries' use, as well as to measure their impact on socio-economic dimensions. The ANCOVA estimates are preferred over the difference-in-differences estimates in scenarios in which the autocorrelation of outcomes is low over time, as it provides a more efficient estimation of the effect (McKenzie, 2012). Autocorrelation tests on residuals for the main outcomes reject the hypothesis of autocorrelation, thus it supports the application of ANCOVA to this analysis. In addition, to capture more clearly the impact of the Fufulso-Sawla road on schooling, sanitation, and health, the regression discontinuity methodology was utilized, both employing the 2012 and 2017 Ghana Living Standards Survey (GLSS) and the primary household data collected. Finally, an interrupted time-series analysis (ITSA) with Newey-West standard errors and one lag (Linden, 2015) was performed to assess the effect of the FSRP on the number of tourists visiting the Mole National Park from 2012 to 2018. The holistic design of the evaluation allows for the assessment of both the direct and long-term impacts of the intervention on households and their communities. Finally, several issues and potential limitations highlighted by the literature and identified during the scoping mission in July 2019 were taken into consideration to ensure unbiased estimates and adequate attribution of outcomes. Furthermore, balanced test outcomes confirmed the comparability of control and treatment groups. In this manner, the design eliminates potential sources of bias both the before-after and simple with-without comparisons have, and is expected to yield an unbiased estimate of the project impact (Annex 2 of the Technical Annexes presents more details on the methodology).

Data and sample: IDEV undertook household and community surveys to generate the required data for this evaluation. The data were collected through tablets and smartphones with the help of enumerators, by using Computer-Aided Personal Interviewing (CAPI), specifically Qualtrics (the digitized data collection tool). Data utilized for the econometric analysis in this study included: primary data collected from 2,393 households (across 17 treatment and 13 control communities): and 150 subjects corresponding to educators, health and tourism professionals. The household and institutional data collection took place simultaneously from 7 to 25 October 2019. Furthermore, the secondary data collection involved an analysis of six household-level national surveys obtained from the Ghana Statistical Services at the enumeration levels and tourism data from Mole National Park authorities.

Evaluation limitations and mitigation strategies:

As with any other evaluation, this evaluation inevitably contains some limitations. The evaluation focused on "micro" impacts that were brought about by the project to households and/or individuals along the road project. Although it is important to examine potential "macro" impacts of the improvement of the road, such as enhancement of overall regional economic activities beyond the areas around the project road, it is beyond the scope of this evaluation. Second, the results of the estimation have some potential limitations highlighted by the literature and identified during the scoping mission. These limitations were taken into consideration to ensure unbiased estimates and adequate attribution of outcomes. These included: (i) the absence of baseline data with the control group; (ii) time inconsistency bias controlled through the use of control variables and panel regressions estimated with state-time fixed effects for time and communities; (iii) self-selection

bias addressed by selecting a counterfactual group with similar characteristics as those in the group treated; and finally (iv) results show only the average effect of all the components of the project and the income declaration in household surveys has its limitations, requiring caution in interpreting income outcomes. The inclusion of other aspects such as a multidimensional poverty index (MPI) allows some triangulation. The evaluation estimates the average impact of all components of the project on development outcomes. The impact of the estimation results reflects the combined impact of all the components of the project. Based on the theory of change (Annex 1 of the Technical Annexes), we categorized outcomes to be examined in this evaluation on key variables of interest (traffic intensity; travel time and transport cost; access to socio-economic services, such as education, health, water supply, markets, and touristic attractions; and long-term outcomes, such as poverty, health, education, and employment). The summary statistics of these outcomes of the treatment and comparison groups at the baseline are presented in Annex 4 of the Technical Annexes. In the next section, we present the estimated impact of the project on these outcomes.



Major Findings/Impacts of the Project

Impacts on transport-related outcomes

Impact on travel time, traffic density, and transport costs

Finding 1: Overall, road construction significantly improved the transportation conditions for the treated population. The commuting time fell by 33 percent in 2019, equivalent to more than 120 minutes per month. Access and frequency of transport also significantly improved. However, the project failed to reduce transport costs.

On average, treated households experienced a 25.3 percent reduction in commuting time in 2015 and a 33 percent reduction in 2019. Before the treatment, the average commuting time was 30 minutes more per trip than the control group. It is worth noting that the three largest commuting saving times correspond to the commute to health-care

facilities, the hospital, and schools. Households living along the Lot 2 section of the road experienced a larger decline than those living along Lot 1. This is not surprising, given that traffic flow on the latter before the road construction was better than in Lot 2. Both males and females seem to have benefited on average equally from the decrease in commuting time.

Figure 10 of Annex 4 of the Technical Annexes presents the commuting time's normal distributions for the five places of reference utilized in the construction of the travel time index. On average, in 2012, households in the treatment group spent about 30 minutes more per trip commuting than the control group. After the construction of the road, however, these same households have a shorter commuting time across locations than the control group. Similarly, the road had improved transport access and frequency.

Fable 2: Daily	<pre>/ traffic flow for f</pre>	top four transport	: modes Fufulso-La	rabanga-Grupe
----------------	---------------------------------	--------------------	--------------------	---------------

Transport mode	nsport mode Baseline 2013		Endlin	e 2016	Observed differences	
	Average daily traffic flows	%	Average daily traffic flows	%	Average daily traffic flows	%
Motor cycles	141.0	26.9	257.3	23.6	116.3	-3.3
Cars (private, taxis)	45.0	9.0	154.2	15.1	109.2	6.1
Pick-ups/vans (pick-ups, land rovers, jeeps, 4WDs)	124.5	28.9	183.5	21.1	59.0	-7.8
Mini buses (seating cap. not more than 25 persons)	21.8	5.7	94.0	10.9	72.2	5.2
Others	131.0	29.5	336.0	29.2	205.0	-0.2
Total	463.3	100.0	1.024.7	100.0	561.3	-

Source: Ablin Consult

Outcomes	2015		2019	
	Average change	Significance	Average change	Significance
Traffic intensity (increase)	121%	•		
Travel time (decrease)	-25.3%	•	-32.5%	•
Transport cost (increase)	14.5%	•	14.5%	•

Table 3: Summary effects and impact of the FSRP – transportation outcomes

Significant effects Or significant effects Significant effects with unexpected sign Source: Evaluation Team

Traffic flows within major towns of project beneficiary districts increased significantly. Average daily traffic flow across three main stations—Fufulso, Larabanga, and Grupe—increased by 121 percentage points, from 463 flows in 2013 to 1,025 flows in 2016. Although motorcycles, cars (both privately owned and taxis), and pick-ups/ vans accounted for most flows before and after the project, the proportion changed. While the proportion of motorcycles declined from 27 percent at baseline to 24 percent in 2016, that for cars increased from 9 percent in 2013 to about 15 percent in 2016 (Table 2).

It is partly attributable to a boost in commercial taxi activities as a result of increased accessibility brought about by the road project. In Larabanga, traffic flow by privately-owned cars and taxis increased by 9 percent over the same period. Triggered by increased access to places previously difficult to access, especially the country's Mole National Park. Table 18 in Annex 4 of the Technical Annexes presents details of the surveys, with emphasis on the top four modes of transportation.

Notably, mini-buses used for commercial purposes saw a marked increase in daily traffic flow across all three stations. In the case of Fufulso, daily traffic flow of mini-buses doubled in 2016, about 10.5 percent, while in the case of Larabanga, average daily flows as of 2016 stood at 7.7 percent, more than three times the situation at baseline.

However, **the project led to a significant increase in transport costs**, by 14.5 percent compared with the control group (Table 3).

Impact on access to socio-economic services

Impact on overall market conditions

Finding 2: The road construction helped significantly improve overall market conditions in treated households by 11 percent in 2015 and 14 percent as of 2019. The presence of ancillary works, in addition to roads, amplified the impact generated by the road. The influence of the ancillary works on economic outcomes, however, is not uniform. Instead, it varies depending on the aspect measured, as well as over time.

The intervention had a statistically significant and positive impact on market conditions along the entire corridor. Its impact lasted and continued to grow four years after its completion. Villages located along Lot 2 experienced a positive impact only after 2015. When schools and markets are added to the road intervention, the market conditions' positive impact is enlarged. For households living in villages removed from the main road by 5 to 10 km, market conditions worsened. Males experienced better market conditions than females. Finally, the lower the level of education of the spouse, the more they saw an improvement in market conditions, albeit the improvement being significantly lower than the one experienced by their husbands or children. The list of measures encompassing the index utilized in the estimation could be found in Annex 5 of the Technical Annexes. Figures 7 and 8 in Annex 4 of the Technical Annexes depict the overall market conditions distribution for the heads of household and spouses in treatment and control groups over time.

On average, the control group faced better conditions than the treatment groups in 2012. Nonetheless, while control group households experienced no change, a 4.4 standard deviation increase was experienced by treatment households, exposing them to better market conditions than the control group in 2019. Spouses experience a 6.28 standard deviation shift.

Impact on overall market integration

One of the goals of the FSRP was to improve the market integration of the Savannah Region with markets in other regions of the country, together with neighboring countries. The variable of interest utilized to evaluate the impact of the road on market integration is an index that measures the geographical reach of households in the acquisition of input products, as well as in the trading of their goods. A description of the index component is found in Annex 5 of the Technical Annexes.

Finding 3: The project improved market integration in the beneficiary areas by a significant 7 percent in 2019. It also fostered market diversification and unleashed synergies with other organizations' initiatives.

The intervention had a statistically significant and positive impact on market integration four years after the construction of the road was completed. Evidence indicates that the FSRP led to a significant increase in the market integration index of 7 percent four years after the road was completed (Table 4). Evidence of synergy with other organizations' work was also identified. Specifically, households that received assistance with business development achieved better integration into other markets than those that did not.

Evidence shows a positive and significant impact of the road and ancillary works on market diversification, an impact of 2.1 percent in 2015, and 2.2 percent four years after completion. As was the case for previous economic outcomes analyzed, findings on the market's diversification lend support for comprehensive interventions rather than singled-focused ones. The most significant increases in trade corresponded to trading with the regional capital (increased by a factor of 3), followed by foreign countries (increased by a factor of 2), and Accra (an increase of 1.5 times). Trading with bordering countries and district capital followed closely with an average increase of a factor of 1.3. Local integration (i.e., trading with nearby and further communities, the district capital, and Accra) oscillated between 0.7 and 1.4 percent. In contrast, trading with foreign countries (i.e., bordering countries, foreign countries, and overseas) oscillated between 0.1 and 0.67 percent.

Rural villages located 5 km from the road transport goods less frequently than those located on the road. As in previous cases, interventions carried out by other institutions also improve outcomes and thus highlight synergies that could be exploited in future interventions. Specifically, households that benefited from post-harvest loss training decreased the frequency with which they transported goods to the market, thus increasing their capacity to ensure stable incomes or exploit better market conditions. Similarly, households that benefited from new technologies reported an increase in the frequency with which they transported goods in 2019. Households with storage facilities transported their goods to the market less frequently.

Also, evidence shows a statistically significant and positive impact of the FSRP on trading conditions. The improvement was generated by road access and by the combination of the road with ancillary works. The latter amplified the capacity of household members to negotiate prices as a result of the road construction. Evidence also suggests that villages removed from the road have less capacity to negotiate prices. Figures 10 and 11 (in Annex 4 of the Technical Report) present the normal distributions for markets' trade diversification index, and the frequency with which households trade in the control and treatment groups. In both cases, significant changes in the indexes' distribution for the treatment group and the shift experience are equal to a one (1) full standard deviation.

Impact on the use of market facilities

Finding 4: The use of markets where facilities were constructed increased in Lot 1, but not in Lot 2. Problems with their location, design, and dusty commuting conditions at the baseline had a negative impact in 2015. The population seems to have adapted to the new locations, however, as by 2019 the negative effect had dissipated.

The constructed market facilities were used in Lot 1 if no problems were reported with the market locations. In markets in Lot 1 at the right location (as per users' needs), the use of the market facilities increased by a statistically significant 4.5 percent. Problems with the location of the markets led to a statistically significant decrease in the market use of 6.2 percent in 2015. By 2019, however, the population seemed to have adapted to the new location, but this was not the case for all markets. Some locations remained unused (Table 8 in Annex 4 of the Technical Annexes). Of the four, only the Sawla market was not used, due to socio-cultural reasons. In the case of Fufulso, it is only used on market days. Similarly, the truck park provided as part of the market facility in Fufulso is seldom used. This is because buses and other vehicles plying the route normally stop alongside the main road and so taxis prefer parking there. Unlike the Sawla market. the situation in the non-use of the Fufulso market and truck park is purely an enforcement problem which the district assembly is best placed to address and resolve.

Households also reported problems with the design of the markets. Nonetheless, this problem did not deter households from using the facilities. On the contrary, in markets where new designs were introduced there was a statistically significant increase in the frequency of market facilities' use in 2015 (6.3 percent increase). However, this impact dissipated by 2019, as users saw it was better to be more visible and display their goods outside the boxes built. Lack of transport access at the baseline and very dusty commuting conditions adversely affected households' use of markets, but this also reverted by 2019, reflecting better transport conditions four years after the completion of the FSRP. Households' sense of responsibility for the maintenance of market facilities significantly increased the frequency of use. However, the impact was relatively small (0.2 percent).

Finding 5: The building of truck parks or the grain storage facility did not affect the frequency of use of these two ancillaries.

No statistically significant evidence was found indicating that the building of truck parks or the grain storage facility affected the frequency of use of these two ancillaries (Table 9 in Annex 4 of the Technical Annexes). Instead, it was the location where the families lived. Households in urban areas or those living closer to the road utilized the truck park more frequently (1.5 and 1.8 percent more frequently, respectively). Among other factors that played a minor but nonetheless statistically significant role in determining the frequency of use of truck parks and grain storage facilities, was the time it took households to reach the market (the closer the commute to the market, the more frequently they used both facilities). Other factors included: (i) the length of time spent collecting water for household use; (ii) the location where households acquire inputs and the types of goods they sold; (iii) transport access at baseline; and (iv) safety and crime recurrence on the road at the baseline. The sizes of the coefficients for these variables were all small.

Impact on the development of new business, quality of employment for households, and economic opportunities for youth

Finding 6: Overall, the FSRP significantly improved the development and arrival of new business (by 12 percent in 2019). However, the impact on the quality of employment of households was only significant in 2015 when the road was associated with schools and health facilities. Also, the impact on economic opportunity for youth was only significant when the road was associated with market facilities.

Outcomes	2015		2019	
	Average	Significance	Average	Significance
Market conditions	11%	•	14%	•
Market integration		•	7%	•
Development/arrival of new business	11%	•	11.6%	•
Type of occupation (road only)	-13%	•	-14%	•
Type of occupation (road + school facilities)	8%	•		•
Type of occupation (road + health facilities)	10%	•		•

Table 4: Summary effects and impact of the FSRP – market conditions outcomes

Significant effects
 Significant effects
 Significant effects with unexpected sign
Source: Evaluation Team

The evaluation found positive results of the development/arrival of new business. The FSRP led to a significant increase in the development and arrival of new business (by 12 percent in 2019). The type of occupation held by three household members was used as a proxy for the quality of employment. Only categories 1 through 10 were included, as these capture jobs carried out outside the home. The higher the value assigned to the type of occupation included in the measurement, the more skilled and the better paid the job typically is. Consequently, higher values of the variable of interest are associated with a better type of occupation. In 2015, treatment groups reported a statistically significant and lower score (-13 percent) in the quality of employment held than households in the control group (Table 4). The same impact was found in 2019. However, the outcome was found not to be even across interventions and years. The inclusion of ancillary works had a positive and statistically significant impact on the type of occupation of households. In 2015, this was the case in villages where schools and health-care facilities (respectively, increases of 8 and 10 percent) were constructed along with the road (Table 4). This is mainly due to the fact that schools and health facilities offered better quality job opportunities, as they required high skills compared with those needed to work as farmers. Also, the new business did generate jobs, including better jobs, but these were located in urban areas. This is why statistically significant and positive results are found for urban areas, both in 2015 (5.9 percent) and in 2019 (5.7 percent).

The variable economic opportunity for youth measures the market conditions for young people's iobs in agriculture and non-farming employment, as well as their access to education, and whether they or their parents could pay for their education. The road helped generate economic opportunities for youth when it was associated with market facilities. This impact persists four years after the completion of the intervention. In 2012, the economic opportunities for youth were poor (1.84 out of 4). Although markets have helped improve them, seven years later, they remain poor (2.17 out of 4) (Table 16, Annex 4 of the Technical Annexes). Additional explanatory variables included in the estimation played a significant role. These include assistance received from other institutions in education, market access, and road access, The latter findings are of interest, as they highlight potential and effective synergies for future projects.

Impact on tourism

Results indicate a statistically significant and positive impact of the road on tourism activity along the road. The increase equaled 1.26 standard deviations in 2015 and the effect continued to increase, albeit slightly, in 2019. The control group shows no statistically significant change. Figure 3 depicts the tourism expansion distribution between 2012 and 2019 for both the treatment and control groups. Findings are consistent with the reality faced by the Damongo-Mole area regarding a shortage



Figure 3: Tourism normal distributions 2012–19, treatment vs control

in hotel infrastructure, as well as the capacity of Mole National Park to handle a significant increase of tourists.

The patronage of tourism among Ghanaians saw a marked increase in the beneficiary districts. The total number of tourists to Mole National Park was estimated at 12.691 tourists in 2005, with the number increasing by an average 384 tourists annually up to 2013. In the first year of the road construction (2013), there was no significant effect on the number of tourists, followed by a significant increase in the annual trend of tourists relative to the pre-intervention trend (Figure 11 in Annex 4 of the Technical Annexes). When the number of tourists is disaggregated by origin (foreign and Ghanaian), the results show that, from road construction, local tourism has gradually substituted tourism of foreign origins. Indeed, the number of foreign tourists dropped considerably from 7,303 in 2012 (before the actual start of road works) to 4,927 tourists in 2015 (end of the works). Meanwhile, the number of Ghanaian tourists went from 7,271 to 12,815 tourists over the same period. This observation is confirmed by the estimations that show that the road construction saw a significant increase in the annual trend of Ghanaian tourists (relative to the pre-intervention trend), by 387 tourists per year and a non-significant negative effect on the number of foreign tourists. Table 17 in Annex 4 of the Technical Annexes shows the results

of the interrupted time series analysis of Ghanaian, foreign, and total (Ghanaian + foreign) tourists in Mole National Park.

Impact on Mole National Park road and the Mognori Bridge

Finding 7: The use of the Mole National Park road and the Mognori Bridge benefited first the communities closer to the road, and by 2019 also villages removed from the main road. Furthermore, the impact goes beyond the immediate beneficiary communities located right before and after the Mognori Bridge.

The use of these two works benefited first those communities that were closer to the road in 2015 (an increase of 3.5 percent). Nonetheless, the benefits reached villages removed from the main road four years later (an increase of 1.9 percent). Specifically, by 2019, there is statistically significant evidence that households in rural communities located between 5 and 10 km from the road increased the frequency with which they used both the Mognori Bridge (1.9 percent higher) and the Mole National Park road (1.7 percent higher use). Furthermore, this also indicates that the impact of the ancillaries went beyond the immediate beneficiary communities located right before and after the Mognori Bridge, or located next to the Mole National Park entrance. There is also evidence that those households that were utilizing the bridge and the Mole National Park road before the FSRP intervention began to be implemented in 2012 increased their use of both ancillaries after their construction (Table 10 in Annex 4 of the Technical Annexes). At a focus group discussion with Mognori and Murigu communities, members agreed that the project, especially with the construction of the bridge, had boosted economic opportunities. They noted that before the construction of the road and bridge, shea nuts picked from the wild were used to just make candle lights, but now women are picking and selling the shea nuts. They added that the use of tricvcles to transport shea nuts to the main market centers had created a new avenue for women to earn higher incomes. In these communities, the charcoal-burning business was booming as the road and bridge made transportation easier.

Impact on access to quality water supply

Finding 8: The project significantly increased households' access to quality water supply by 17 percent, though bad relations between staff (teachers and nurses) and communities reduced the impact.

The FSRP had a positive and statistically significant impact on households' access to quality water supply,8 at about 17 percent as of 2015 (Table 5). The establishment of the water treatment plant as part of the project is largely responsible for the impact observed. This saw the connection of more than 10,000 people within the Fufulso catchments to potable water sources. In some communities, the provision of boreholes alongside health facilities and schools also contributed to increasing households' access to drinking water. For example, health-care facilities also contributed toward improving the water quality by 7 percent, but the impact was not statistically significant until 2019. A combination of factors explains the low impact of the mentioned ancillaries. Findings in this study highlighted the poor conditions encountered in some of the markets' and schools' boreholes.

Furthermore, relational factors affected access to clean water collected through boreholes. Typically, women reported that if they had problems accessing the boreholes and/or if they had a bad relationship with school or health staff, they generally chose not to use the boreholes. Instead, they collected water from the river or creeks. Households living close to the road and those that had access to schools face fewer water-quality problems. In the case of the water treatment plant, qualitative interviews revealed that under-utilization of the plant (about 8 percent) to an extent affects the revenue generation, which in turn undermines the maintenance of the plant to quarantee reliable water supply. Also, field visits to the water treatment plant revealed that, due to poor maintenance, some installations were subject to rusting and had the potential to adversely affect water quality.

Impact on the overall sanitation conditions

Finding 9: Overall, the project generated a statistically significant and positive impact on households' access to and use of improved sanitation.

Findings support a positive and statistically significant impact of the road and ancillary works on sanitation conditions. The construction of the road, combined with improved access to health facilities, helped to improve access to sanitation services by 11 percent in 2015 and 14 percent in 2019 compared with control households (Table 5). Two main components of the sanitation ancillary works were the construction of boreholes and Kumasi Ventilated Improved Pits (KVIPs), which is an improved version of the standard latrines. All communities along the road had boreholes and KVIPs installed. Thus, a sanitation use index was constructed based on households and staff reporting on the frequency with which they used these two ancillary components. The higher the value of the index, the more frequently were the boreholes and KVIPs used by the subjects. In some communities, these sanitation facilities were attached to markets, schools, and health-care facilities; in others, they were not.

Outcomes	2015		2019	
	Average Change	Significance	Average Change	Significance
Quality water supply Index	17%	•		•
Overall sanitation conditions Index	10.8%	•	14.4%	•
Significant effects No significant effects Source: Evaluation Team				

 Table 5:
 Summary effects and impact of the FSRP – socio-economic services

Relational factors also affect outcomes. Specifically, the quality of the relations between medical staff and community members, as well as the quality of the relationships between school staff and community members, when positive, help to increase sanitation conditions. Poor quality relations deter people from using the sanitation facilities attached to the school and health-care facilities.

Socio-economic impacts of the project

Impact on household incomes

Finding 10: Overall, the FSRP improved the treatment households' income significantly in 2019. Evidence of a more positive impact on household incomes was found in communities where the road construction was accompanied by health-care facilities.

A positive and statistically significant impact of the road was found on household incomes in 2019, though no statistically significant difference was found between the treatment and control groups as a consequence of the road construction in 2015. The project had significant positive effects on household incomes in 2019, with an increase of

US\$15 per week⁹ or an increase of 84 percent from the average income in 2012 of US\$18 (Table 6). The impact was not uniform across ancillary interventions. Evidence of a more positive impact on household incomes was found in communities where the road construction was accompanied by health-care facilities (an increase of US\$18, or an increase of 100 percent). As expected, communities closer to the road reported higher levels of income, but the increase was not statistically significant.

As time passed, the use of sanitation facilities increased by 2.5 percent, especially among families living in communities in which health-care centres were built. Other factors influencing the use included: the location of sanitation facilities in Lot 2 (it increased by 2.7 percent in 2019); the time spent to collect water at the baseline (if more than 120 minutes, this increased by 1 percent); commuting security factors on the way to or from schools and the condition of the facilities (children's use of KIVPs increased by 6 percent in 2015, but the impact disappeared in 2019 due to poor maintenance in schools); and interpersonal relational factors, such as the quality of the household relationships with school staff. and their sense of ownership and responsibility regarding boreholes built in their communities.

Table 6: Summary effects and impact of the FSRP – household incomes

Outcomes	2015		2019	
	Average Change	Significance	Average Change	Significance
Household incomes		•	84% (US\$15)	٠
Caralitanata (Kasta 🖉 Maninatianata (Kasta 🦛 Caralitanata (Kasta a))				

Significant effects Volignificant effects Significant effects with unexpected sign Source: Evaluation Team

Figure 3 of Annex 4 of the Technical Annexes depicts the frequency of use of the two sanitation components, KVIPs, and boreholes. The evaluation also found uneven use among borehole users. Because women typically collect the water, they used boreholes, on average, more frequently than other household members. The boreholes of their choice were the community ones over those attached to other ancillary facilities. The use of the latter had decreased between 2012 and 2019 (Figure 4 in Annex 4 of the Technical Annexes).

Impact on health outcomes

Finding 11: There was a positive and significant impact of the road on the overall health of beneficiaries (8 percent increase in 2015 and 9 percent in 2019) everywhere due to increased access to health facilities. The size of the road's treatment effects on access to health facilities was an 11 percent increase in 2015 compared with control households. The impact continued to expand four years after the intervention was completed (13 percent in 2019). Households also report being able to secure more skilled medical personnel.

Econometric analysis based on the primary data collected finds a robust, positive, and statistically significant impact of the FSRP on the health conditions of the population living in the project area of influence. Table 18 in Annex 4 of the Technical Annexes reports the outcomes of the impact estimations on the ability of household members to access health services. The index captures the ability of household members to reach clinics, use of ambulance services, secure medical personnel for the clinics, and the frequency with which they need to be referred to the main hospital. The results show that the road led to an increase in access to health facilities by 11 percent in 2015 (Table 7). In villages where health clinic facilities were built, the impact was amplified (around 3 percent additional impact). The impact continued to expand four years after the intervention was completed (13 percent in 2019). The villages removed from the road do not seem to reap the same benefits. A positive and statistically significant impact of the road on the overall health of beneficiaries (8 percent increase in 2015 and 9 percent in 2019) was found also. No statistical evidence of the construction of the road was found on child mortality or prenatal care. apart from households alongside the road for which a significant increase (41 percent) was observed in terms of prenatal care. This impact resulted from the increase in the frequency of households' use of health-care facilities.

Finding 12: By 2019, the introduction of the road in conjunction with health-care facilities has led to a statistically significant increase in the frequency of households' use of health-care facilities.

No significant treatment effects were found in 2015 on the use of the health-care facilities. However, in 2019, the use of health-care facilities increased by 4.8 percent (Table 7 in Annex 4 of the Technical Annexes). The presence of schools' ancillaries also helped increase the frequency with which health-care facilities were

Table 7: Summary effects and impact of the FSRP – socio-economic outcomes

Outcomes	2015		2019	
	Average Change	Significance	Average Change	Significance
Overall health	8%	•	9%	•
Prenatal care for HH (alongside the road)	41%	•		•
Use of health facilities		•	4%	•
Access to healthcare	11%	٠	13%	۲

Significant effects No significant effects Significant effects with unexpected sign Source: Evaluation Team

used by 2 percent. Albeit small, the impact signals synergies generated between the road and the ancillary works. The evaluation also found a 2 percent increase in the frequency with which health-care facilities were used by households living 10 km removed from the road. Findings were consistent with the previous one in the literature, which highlighted the generation of the positive spillover effect of main roads on ancillary roads' access. This access, in turn, facilitated the population's capacity to access health services [Egan et al. (2009), Elvik (2009)]. Also, having a sense of ownership and responsibility for maintenance of the health facility increases the frequency with which the services were used. but 56 percent of households reported never being asked for assistance. Assistance was only requested when their livestock invade health-care premises, which is widespread and of frequent recurrence (97 percent of the people reported this happening).

Finding 13: Relational factors were also important for securing improved personal health and services. The quality of family relations, as well as the existence of a good rapport between medical staff, community, local officials, and medical supervisors significantly affected outcomes.

Relational factors also influenced the outcomes in this case. Having a good relationship with the medical personnel at the clinic increases access to health care. For women in a monogamous marriage, access to health care also increased vis-à-vis those women who were living under alternative marital arrangements.

A positive and statistically significant impact of the road was found on the overall health of households. Here the variable of interest measures the reported overall quality of health of the heads of household, spouses, and children. Once again, relational explanatory variables were also found to influence outcomes. The better the quality of the relations of the household members, the better was their overall health. Similar results were found when examining the recurrence of illness, but a higher incidence of illnesses among treatment households vis-à-vis control households was found (Table 12 in Annex 4 of the Technical Annexes).

There is also statistically significant evidence that the road and health-care facilities built contributed to securing medical personnel to run the health-care facilities and securing, on average, better qualified medical personnel. Good relations between the community and the medical personnel at the clinics helped secure more qualified personnel. Indeed, when analyzing the reasons that motivated health-care professionals to work along the FSRP area, good relations with the community, with local government officials, and with supervisors appear as one of the drivers of staff's decisions to take the job in 2015. Synergies with interventions of other institutions were also found here.

Villages with access to housing and clean water through boreholes amplified the positive impact on health achieved by the road. Specifically, results indicate that villages where households had received help from other institutions to secure clean water, were able to secure qualified personnel. The finding lends support for the addition of boreholes to the clinics and other areas in the community.

Consistent with the findings regarding the reduction of illness, no statistical evidence is found of a positive impact on nutrition being generated by the inception of the road or the new health-care facilities, except for villages located in Lot 2; a positive and statistically significant impact on nutrition is found for villages located along this section of the road. Consistent with previous findings, male-headed households have a higher nutritional intake than females.

Treatment households in 2012 were worse off than control households on all three counts: access, overall health, and securing quality medical personal. However, the new road helped households to catch up or even bypass the health conditions enjoyed by control group households by 2015. The mean assessment moved from 2.0 for each index component to 3.13. Figure 15 of Annex 4 of the Technical Annexes depicts the normal distribution of access to health-care services, overall health, and medical personnel qualifications between 2012 and 2019, both for the treatment and control households.

Impact on education outcomes

Finding 14: The construction of the road and the educational facilities together had a positive and statistically significant impact on education.

There is a statistically significant and positive impact of the road on educational achievement. Subjects who benefited from the new road when they were in primary school completed 5.5 grades more than those who did not. If they were living in urban areas, they completed 11 years more of education than those who lived in rural areas. Demographic characteristics also influenced outcomes. Subjects who report belonging to Muslim families and parents who are farmers achieved lower educational levels.

Similar results were seen when estimating the primary school completion rate. Subjects who benefited from the road while they were in primary school, report a 25 percent higher probability of completing primary school than those who did not. The road also had a positive statistically significant impact on school attendance. Children who were at school-age when the road was constructed had a 3.96 times higher probability of attending school. If they lived in rural areas, removed from the road, the probability of them attending school decreased. Finally, males had a higher probability of attending school than females.

Children in the treated group had a lower frequency occurrence of dropping out than children in the control group. Having good family relations also reduced the school dropout rate. No evidence was found of any impact of the road on literacy rates.

Impacts on poverty reduction

Finding 15: Multidimensional poverty was significantly reduced in the treatment group because of the road's construction. Combining the road with schools and markets further contributed to an increase of the overall impact on multidimensional poverty by 20 and 15 percent, respectively, in 2015.

Construction of the road generated a positive impact on poverty reduction among beneficiary households. The multidimensional poverty index (MPI) decreased significantly among beneficiary households, by 2.16 percent in 2015 and 2.59 percent in 2019 (Table 11 in Annex 4 of the Technical Annexes). Thus, the project impacted positively not only on the living standards of beneficiaries but also on education and health outcomes.

The ancillary works generated a proportionately higher additional effect than their additional costs.

The road construction alone was estimated to have led to a statistically significant decrease in MPI among beneficiary households by 2.16 percent in 2015. Households that benefited from a school in addition to the road experienced a significant additional reduction in the MPI of 0.54 percent, increasing the total decrease in MPI (road and school) to 2.70 percent. The additional effect of the school represented 20 percent of this overall effect. The construction of a market in addition to the road led to a statistically significant reduction in MPI of 0.39 percent and a total reduction (road and market) in MPI of 2.55 percent. The additional effect of the market represented 15 percent of this total effect. The ancillary works made it possible to generate a proportionately higher additional effect on the MPI than their additional cost. In 2019, only the markets led to a significant further reduction of MPI by 0.35 percent, which brings the total effect of road and markets to 2.94 percent. The additional effect of markets in 2019 represents 12 percent of this overall effect. Outcomes are consistent with previous findings in the literature, such as Aktin and Donaldson (2015), and provide information regarding the type of factors that can impact positively the primary intervention, as recommended by Van de Walle (2009).

Table 8: Summary effects and impact of the FSRP – poverty re	eduction	outcomes
--	----------	----------

Outcomes	2015		2019	
	Average Change	Significance	Average Change	Significance
Multidimensional Poverty Index	-2.1%	•	-2.6%	٠
Significant effects No significant effects Significant effects with unexpected sign				

Source: Evaluation Team

Impacts on institutional networks and relationships

Finding 16: Both the road and ancillaries had a significant positive impact on the institutional networks along the corridor. Ancillary works were effective channels of these relations, as they provided a physical place of encounter and to share information.

Four groups of measures are included in this institutional analysis section. These were: the quality of life index, the social responsibility index, the civic responsibility index, and the personal and social ills index. Three of them measure life in three fundamental pillars of any society: family, civic life, and social networks of solidarity. They are all important for the development of functional markets and investment, and the stability and health of social life in general. The personal and social ills index facilitates the identification of fundamental problems in society. It also highlights possible lines of work for future interventions.

Impact on the quality of life

The road and ancillary works contributed to improving the quality of life of household members, with the impact lasting over time. Evidence shows a positive and statistically significant impact (3 percent) for head of household and child, as well as 2 percent for the spouse, in 2015. This improvement lasts only for the head of household, with a positive and statistically significant increase of 4 percent four years after the road completion (2019). School interventions emerged again as a point of encounter for all members of the household. Households in villages removed from the road by less than 5 km also showed a positive impact on the quality of their lives. The benefit does not reach those further removed. The impact on medical and school staff is significantly lower.

Impact on social responsibility

Similarly, a positive statistically significant treatment effect of the road on the living of social responsibility was found for staff and spouses. In 2015, social responsibility increased by 27 percent for staff and 1 percent for the spouse. Ancillary works amplified the impact for the heads of household and children, in some cases. The impact on each of the subjects varied, however. The diverse kinds of ancillary works integrated into the road project provided different opportunities for the population to come together depending on their circumstances and professions. Thus, while for the head of household, schools, markets, and truck parks serve as channels for subjects to live out their social responsibilities, for the spouses, the majority of which are women, the optimal space seems to have been schools. Along the same lines, while the road did not seem to influence the way youth lived their social responsibility, schools, where they spent long hours, did. Outcomes suggest a different type of pattern for medical and school staff. While schools seem to have been a positive channel for living out their social responsibility, health-care facilities did not seem to be. Neither were markets. All in all, the multi-dimensional design of the intervention seems to have stimulated the living of social responsibility in the communities.

Impact on civil responsibility

The overall impact of the FSRP and its ancillary works on the living of civil responsibilities is weaker. For the heads of household and children, the positive, statistically significant effect of the intervention was felt with a lag (respectively, 0.3 percent). For the staff, evidence of an impact (20 percent increase) is found in 2015.

Impact on personal and social ills

For heads of households, markets are a path of exposure to social ills, while truck parks and health centers seem to reduce this exposure. At the same time, women face social ills along the road and in the market. They reported facing social and personal ills along the road (specifically personal safety and crime), conditions that worsened after road construction by 47 percent, representing a shift in the evaluation of conditions from poor to deplorable (Table 16 in Annex 4 of the Technical Annexes). This is consistent with previous findings reported in the analysis, as it indicates that spouses living in remote villages have lower exposure to social ills. On the other hand, children are exposed to school. They spend several hours there and are more likely to experience or hear about these types of problems while at school and/or conversing with their friends.

Unintended impacts of project interventions

As part of the evaluation, unintended positive and negative effects were identified.

Finding 17: The project had positive and negative unintended impacts. On the positive side, the project promoted greater social cohesion among beneficiary communities, reduced the incidence of highway robbery, and helped to catalyze investments from other development partners in support of water supply. On the negative side, the project contributed to the degradation of the environment through growth in charcoal burning activity in the three beneficiary districts.

The FSRP promoted greater social cohesion among beneficiary communities and reduced the incidence of highway robbery. The equitable distribution of the ancillary facilities brought greater social cohesion and peace among communities along the corridor of the road. Focus group discussions revealed that inter-community conflicts decreased because communities depend on each other for varying social services. In Sawla, for instance, a key informant indicated that:

⁶⁴...misunderstanding between two beneficiary communities nearly resulted in conflict but for the sake of the equitable distribution of ancillary works, they resolved their differences amicably. This was because one community benefited from a health center while the other benefited from a school. Hence, this inter-dependence quelled the impending conflict.⁹⁹

In terms of security, the project contributed to a reduction in incidents on the highway. Concerning robbery, an official at the Central Gonja District Assembly had this to say:

⁶⁶ Before the road, armed robbery activity was high especially during the rainy seasons as the bad nature of the road slowed down the movement of cars. But this has reduced significantly as a result of the road.⁹⁷

Furthermore, the FSRP helped to catalyze investments from other development partners in building on its outputs and outcomes. The World Bank is expanding the network of the water treatment plant—one of the many ancillary interventions supported under the FSRP—with a sum of US\$1 million to address plant maintenance challenges, as well as increase its operating capacity for increased coverage to other households.

In terms of negative unintended consequences, qualitative interviews point to an impact on environmental degradation. The project is adversely affecting the environment due to a marked growth in charcoal-burning activities in the three beneficiary districts. At the same, time, however, district assemblies in the three districts are earning more revenues from the booming charcoalburning business. Also, the construction of the road facilitated Rosewood logging within the project's enclave—a phenomenon that continues to threaten the Mole National Park in particular.

Sustainability of the project benefits

As shown above, there is statistical evidence that the FSRP did indeed improve most of the livelihood indicators of the beneficiaries' communities. But, are these benefits sustainable in the long term? This section explores the extent of the sustainability of the benefits of the road project. The evaluation points to the fact that the FSRP had a large impact on the beneficiaries' communities. The sustainability of the observed impact is however unlikely as the continued deterioration of the road (Lot 2) and some related ancillaries such as schools and health-care facilities may ieopardize the project. This is mainly due to: (i) irregularities with the construction of road infrastructure and ancillary works; (ii) insufficient ownership on the part of the population resulting from the absence of effective participation during the implementation process; and (iii) inappropriate maintenance of the road and its related ancillaries.

Irregularities with the construction of road infrastructure and ancillary works

Finding 18: The project experienced irregularities with the construction of road infrastructure and ancillary works. There was widespread awareness of these problems among project stakeholders. There is little evidence, however, that issues of maintenance were addressed during the construction process in Lot 2 or afterward. Similarly, little support is found for communities acknowledging responsibility for the ancillary works in their communities.

Overall, outcomes indicate widespread awareness of problems with the construction of both the road and the ancillary works. This was especially the case in Lot 2, where households reported irregularities during the construction process. There is also evidence that problems were discussed at the community level, as well as raised by local authorities and the construction companies/ consultant. The former assisted in resolving ancillary construction problems in Lot 1, but not in Lot 2. There is little evidence, however, that issues of maintenance were addressed during the construction process in Lot 2 or afterward. Similarly, little support is found for communities acknowledging responsibility for the ancillary works in their communities.

By 2015, the probability of households reporting problems with the road increased from 58 to 88 percent. This was especially the case in Lot 2 (16 percent higher probability). Outcomes indicate that issues were raised with chiefs, government officials, and construction companies/consultants. Four years after, problems with the road continued to be reported along Lot 2 and by households that benefited from the Mognori bridge. Markets were still an important institutional channel of communication and a source of shared information. Households linked road deterioration problems with irregularities during the construction stage of the FSRP. Similar outcomes were found when analyzing staff responses.

There is no evidence of any discussions taking place regarding the maintenance of the different facilities. This was the case before, during, and after the construction. An exception was the Mognori bridge. Statistical findings were consistent with what was encountered during the exploratory trip. Specifically, during the meeting with district assembly members, they raised concerns with the fact that they did not have maintenance plans for the ancillary works constructed in their district. Estimation outcomes regarding the significance of household characteristics suggest that awareness of construction problems spread among all household members independently of those households' economic conditions.

Ancillary problems reported by medical and school staff vary with the type of ancillary facility. Health clinics along Lot 2 reported the highest probability of having problems in 2015 and thereafter (50 and

80 percent in 2015 and 2019, respectively). Staff also reported problems with construction appearing during the construction (86 percent) or the first three years after the construction (12 percent) (Table 1, Annex 4 of the Technical Annexes).

Lack of sense of ownership

Finding 19: Efforts to engage the community remained at the level of information but not participation. This was the case for communities where health facilities were built or refurbished.

The absence of participation during the implementation process compromised the sustainability of the project, and also reduced the positive initial impact four years after the completion of the road and ancillaries. Variables utilized to measure engagement include knowledge of the road construction project before the construction began, as well as of meetings organized and the degree of participation on the part of the different community constituencies.

The majority of spouses (58 percent) and children (56 percent) heard about FSRP before its construction, while the proportion of husbands and staff reporting knowledge of the project before its construction began was lower. Family and civic networks such as district leaders and chiefs were typically the sources of this information. Medical and school staff instead learned about the FSRP through schools, NGOs, or the construction companies directly. Except for the children, awareness of meetings organized for the FSRP was low among the population (heads of households at 24 percent. spouses at 14 percent, and staff at 14 percent). and so was the percentage of beneficiaries invited to participate in them. This was the case for women and children. However, the majority of those given the opportunity to participate in the meeting reported a high level of participation in them, at over 63 percent (Figure 2, Annex 4 of the Technical Annexes). Those who attended the meetings found them to be informative, but they reported little willingness on the part of organizers to answer questions or receive input.

The main findings reported in Table 1 of Annex 4 of the Technical Annexes suggest that efforts to engage the community remained at the level of information, but not participation. This was the case for communities where health-care facilities were built or refurbished. Finally, findings on the regressions for medical and school staff reflected the same dynamics. They typically learned about the FSRP project through schools, community chiefs, and district officials but not through the AfDB. Staff report a 73 percent higher probability of organizers being open to input if community leaders attended the meetings. Outcomes from the estimations carried out on staff produced similar results.

Inadequate maintenance and management

Finding 20: Regular maintenance is a critical precondition for sustaining the positive impacts that road and ancillary facilities bring to communities. Under this project, however, the field visit and interviews with beneficiaries revealed inadequate maintenance as a recurring theme across many of the facilities provided, especially for health centers, schools, and bungalows built for teachers and nurses.

The project led to significant improvement in school facilities for treated communities in 2015, but the improvement did not last. By 2019 school conditions significantly deteriorated in treatment communities to the point of reversing the variables' coefficients' sign as shown in Table 4 of Annex 4 of the Technical Annexes. We also find that, on average, treatment communities that received schools in Lot 1 reported a statistically significant improvement in school conditions in 2015 (9.2 percent), while those in Lot 2 reported significant deterioration.

No statistically significant difference was found in the conditions of health-care facilities between the treatment and control groups. The best conditions were noted at the facilities' baseline (2012), but if problems with them were reported during the construction then the conditions of the facilities deteriorated by 2015 and further worsened by 2019. Another factor contributing to the deterioration of the health-care facilities' physical condition was the absence of training before staff received the facilities. Motivational factors also influenced the maintenance of the health facilities. Lack of staff motivation generated by poor relations with colleagues worsened the physical conditions of the facilities. This was especially the case along Lot 2. Similarly, staff who reported being motivated by the housing provided to health-care personnel reported better facility conditions (Table 4, in Annex 4 of the Technical Annexes). At the Fufulso Health Center, officials asserted that, while construction of the dispensary had helped improve health-care services at the center:

⁶⁶...all solar systems installed for bungalows for nurses are down. Also, sinks in the health facility are not working. This has been reported to the Ghana Health Services, but nothing has been done as yet.⁷⁷ Also, solar panels installed to power some health centers were either burgled or were not fully functional. For the water treatment plant, no maintenance plan was in place because of the under-utilization of the plant. This problem was further aggravated by a lack of maintenance plans at the local government authority level. For the road, the Larabanga-Sawla section was degraded even before its delivery, reducing the impact of the project. Several reasons contributed to this: poor scheme design, weak organizational and institutional capacity, and the lack of active community and government ownership of the project.

Interpersonal relational factors affected the physical conditions of school facilities. If the staff was motivated by good relationships at their school, by their colleagues, as well as by their community and authorities, then the physical condition of their school was better in 2015 and 2019 (Table 3 in Annex 4 of the Technical Annexes).





Conclusions, Lessons, and Recommendations

Conclusions

The impact evaluation of the Fufulso-Sawla Road Project demonstrated the transformative nature and multidimensional impact of combining road, social and economic infrastructure in a single intervention on poverty reduction. It also highlighted the importance of institutional and relational factors in obtaining, as well as ensuring, that the impact generated lasts. Finally, it provided robust evidence for the effectiveness of implementing comprehensive and integrated interventions over single-focused ones, as it amplified the project's impact and generated synergies that contributed to accelerating development. However, the impacts observed were not uniform but varied depending on the economic or social aspect being measured.

Subjects included in the sample are representative of the population in the region of intervention, and the control group selected is comparable to the impacted population overall. Therefore, findings are not only of interest for the communities surrounding the two roads included in the study but could be of assistance for current and future development efforts in all three districts included, and for the Savannah Region of Ghana as a whole.

Through the impact evaluation of the FSRP, the Bank sought to answer a set of questions. The main findings are as follows:

The road project improved significantly the transportation conditions in beneficiary communities. Commuting time was reduced by over two hours, and average daily traffic flow across the three main stations-Fufulso, Larabanga, and Grupe-in the project area increased. However, the project led to a significant increase in transport costs by 14.5 percent compared with the control group.

The road project interventions improved market conditions. favored market integration and diversification, and led to the development/arrival of new business. Market conditions, including households' access to both farming and non-farming opportunities, inputs of production, and access to agricultural and other skills training, increased by 14 percent in 2019 as a result of the road project. The FSRP contributed to a significant increase in the market integration index and market diversification. Finally, the project led to a significant increase in the development and arrival of new business. However, the impact on employment opportunities for both adults (type of occupation) and youth (economic opportunities) was only significant when the road was associated with health facilities.

- Household incomes significantly increased. However, the impact was not uniform across ancillary interventions. Evidence of a more positive impact on household incomes was found in communities where the road construction was accompanied by health-care facilities. As expected, communities closer to the road reported higher levels of income, but the increase was not statistically significant.
- The project improved the living conditions of the beneficiaries. The project significantly improved households' access to a quality water supply and sanitation services. These improvements are partly accounted for by the construction of the water treatment plant, the

sanitation facilities, and the boreholes associated with schools and health facilities. Also, the project improved significantly the overall health conditions of the beneficiaries. However, there was no statistical evidence of positive effects from the project on child mortality or prenatal care, apart from households alongside the road for which a significant increase (41 percent) was observed in terms of prenatal care. Finally, concerning the education outcomes, the evaluation found statistically significant positive effects on school attendance and completion.

- The evaluation demonstrated that the Banksupported integrated road project had the desired effects on poverty reduction. Multidimensional poverty, which considers that poverty reduction can come from changes in sources of income and better access to social services, was significantly reduced as a consequence of the integrated road project. This result shows the project had significant positive effects not only on the living standards of beneficiaries but also on education and health outcomes. In communities where road construction was accompanied by schools and market centers, impacts on household poverty reduction amplified. This underscores the significance of adopting integrative approaches to road projects, especially when targeting poor and economically disadvantaged areas.
- The ancillary works generated a proportionately greater additional effect than their additional costs. Households that benefited from a school or a market in addition to the road experienced a significant additional reduction in the MPI, one that was higher than the total cost of the ancillary works.
- The intervention also benefited girls and women, albeit to a lesser degree than men. Indeed, there is evidence of males using the road more than females, but females with higher levels of education also used the main road more frequently. In terms of poverty reduction, the impact

of the project on MPI is greater for male- than for female-headed households. Males also benefited more than females from the positive impact on education generated by the road. However, female school enrolment and retention increased, maternal deaths decreased, access by women to local markets and water collection improved, and their economic opportunities expanded.

- I The project had positive and negative unintended impacts. Regarding the positive unintended consequences, the project promoted greater social cohesion among beneficiary communities, reduced the incidence of highway robbery, and helped to catalyze investments from other development partners in support of water supply. However, on the negative side, the project contributed to the degradation of the environment through growth in charcoal-burning activities in the three beneficiary districts.
- The sustainability of development outcomes of the Bank-supported road project and related ancillary works was highly unlikely. Regular maintenance is a critical precondition for sustaining the positive impacts that road and ancillary facilities bring to communities. Under this project, however, the field visit and interviews with beneficiaries revealed poor maintenance as a recurring theme across many of the facilities provided, especially for health centers, schools, and bungalows built for teachers and nurses. For instance, solar panels installed to power some health centers were either burgled or were not fully functional. For the water treatment plant. no maintenance plan was in place because of the under-utilization of the plant. This problem was further aggravated by a lack of maintenance plans at the local government authority level. For the road, the Larabanga-Sawla section was degraded even before its delivery, reducing the impact of the project. Several reasons contributed to this: poor scheme design, weak organizational and institutional capacity, and the lack of active community and government ownership of the project.

Lessons

The following are the key lessons from this impact evaluation.

- 1. Integrating community development interventions into road transport infrastructure brings an added value to, and fast tracks, the Bank's poverty reduction, job creation, and inclusive development efforts. Results from the evaluation showed that combining road construction with the construction of market centers and schools in a largely rural context yielded additional positive impacts on multidimensional poverty reduction. This underscores the need to pay close attention to the context when designing such integrated road interventions.
- 2. Relational factors are key to the sustainability of integrated road interventions similar to the FSRP, as they can amplify or jeopardize a project's impact. For example, failure to ensure active participation of project beneficiaries beyond 'information sharing' undermined the maintenance of ancillary facilities and thus affected sustainability. Conversely, areater multisector collaboration demonstrated among satff of the Bank's Ghana country offices contributed to the success of the project by drawing on availability of the right caliber and mix of staff. This lends credence to the important role of the Bank's Development and Business Delivery Model (DBDM) which, among others, underscores the importance of having the full complement of staff to support its operations in Regional Member Countries.
- 3. Coupling road projects with community development interventions can increase the likelihood of benefits accruing to women and girls. In particular, the evaluation demonstrated marked gains regarding women's and girls' access to social services such as health and education. This form of integrated intervention can add a layer of gender sensitivity to road interventions.

Recommendations

IDEV makes the following recommendations:

1. Enhance the Bank's integrated approach to its road investments to foster development impact in terms of poverty reduction. Evaluation results showed that a road alone is not enough to tackle poverty. The poor, lacking assets to take advantage of better opportunities that a road may bring, benefit from additional support. This suggests that integrated projects are necessary to tackle poverty effectively. Indeed, while the inclusion of community development projects in road projects such as the FSRP make up a small fraction of overall cost, they can make a significant difference, particularly for increasing access to social services. Such projects provide an impetus for amplifying the social impacts of roads, which in the long run brings an added value to multidimensional poverty reduction and inclusive development. Thus, first, the Bank should explore more of these integrated road projects and proactively adopt them as flagships for its inclusive growth, poverty reduction, and rural development efforts. Second, it should, on the back of its One-Bank approach, step up support for community development components of road projects by deliberately committing more financial resources to them and not treating them as add-ons or afterthoughts. However, the Bank should also be cognizant of the fact that such approaches may not work in all contexts and, hence, should adapt the choice of the ancillary works/services to local realities.

2. Improve the quality of road projects' design and results focus. Given the multidimensional nature of such interventions, having a theory of change that demonstrates causal relationships and impact linkages is critical. This will facilitate the design, implementation, and evaluation of the interventions. The Ghana FSRP used an integrated approach. However, the project documents did not clearly present how the road will interact with the ancillary works to achieve the expected intermediate outcomes. Therefore, for similar future investment: (i) the Bank should base its integrated road intervention design on an evidence-based well-articulated theory of change, with a clear pathway through which the logic model would occur; and (ii) the baseline for the treatment and control groups should be established during the design phase of the intervention for such large scale, innovative, and flagship projects to determine not only whether an intervention is effective, but also to compare options for making interventions more effective.

3. Strengthen the human and institutional capacity to sustain development gains. Neglect of periodic maintenance threatens

the long-term social and economic benefits from the road and ancillary facilities. The Bank should strengthen the human and institutional capacity needed for the sustainability of road projects and their ancillary facilities by: (i) engaging in dialogue with the government to explore partnership-based approaches where a memorandum of understanding can be signed with local government authorities on maintaining the facilities; (ii) encouraging meaningful beneficiary participation and contribution in all infrastructure maintenance activities, including labor and material inputs, thus creating ownership through risk-sharing, to guarantee the sustainability of actions; and (iii) adequately mitigating or minimizing unintended environmental degradation.





Annexes

Annexes

The "Technical Annexes" document can be found on the following page: <u>http://idev.afdb.org/en/document/evaluation-transport-impact</u>

List of annexes:

- 1. Project Logic Model and Theory of Change
- 2. Methodological Approaches
- 3. Impact Analysis Design Econometric Approach
- 4. Some Empirical Results
- 5. Indexes' Definitions by Area Evaluated
- 6. Maps on Treatment and Control Roads, and Ghana Road Plan 2015–2035

References

Ablin Consult. 2015. Final Midterm Report: Vol I & II, Consulting Services for Impact, Monitoring, and Evaluation for the Fufulso-Sawla Road project, Ghana. Adukia. Aniali. 2016. "Sanitation and Education." Applied Economics. 9(2), 23-59.

Adukia, Anjali, Sam Asher, and Paul Novosad. 2017. "Educational Investment Responses to Economic Opportunity: Evidence from Indian Road Construction," *Becker Friedman Institute for Research in Economics Working Paper No. 2017–08.*

African Development Bank (AfDB). 2015 "Project Completion Report for Public Sector Operations (PCR)", November.

African Development Bank (AfDB). 2014–16. Transport Operation Missions: Aide Memoire, March 2014, October-November 2014, April 2015, April 2016, and November-December 2016.

African Development Bank (AfDB). 2017. Ghana-Fufuldo-Sawla Road Project, Project Completion Report, African Development Bank, December.

- African Development Bank (AfDB). 2019a. "Terms of Reference: Impact Evaluation of AfDB-funded Ghana Fufulso-Sawla Road Project", Inter-Office Memorandum Independent Development Evaluation, IOM/DEV/JM/IE/2019/06/01.
- African Development Bank (AfDB). 2019b. "Impact Evaluation of the AfDB-Funded Fufulso-Sawla Road project in Ghana: Scoping Mission From 1–8 July 2019), Inter-Office Memorandum Independent Development Evaluation, Back to Office Report (BTOR) IOM/BDEV/JW/IE/2019/07/01.
- Aguirre, Maria Sophia. 2011. "Achieving Sustainable Development: An Integral Approach to an Economics Perspective", in *The Ethics of Sustainable Development*, Luis G. Franceschi), Strathmore University Press.
- Aguirre, Maria Sophia. 2013. "An Integral Approach to an Economic Perspective: Consequences for Measuring Impact," Journal of Market and Morality, 16(1), 53-67.
- Aguirre, Maria Sophia. 2018. "Inequality and Growth: Exploring a Relational Dimension", in Daniela Ortiz and Martin Schlageds. Poverty, Inequality, and Injustice. London: Springer.

Alesina, Alberto, and Paola Giuliano. 2016. "Culture and Institutions", Journal of Economic Literature, 53(4), 894-944.

Atkin, David, and Dave Donaldson. 2015. "Who's Getting Globalized? The Size and Nature of Intranational Trade Costs," NBER Working Paper No. 21439.

Atkin, David. 2016. "Endogenous Skill Acquisition and Export Manufacturing in Mexico," American Economic Review, 106(8), 2046–2085.

Asher, Sam, and Paul Novosad. 2019. "Rural Roads and Local Economic Development", American Economic Review, Forthcoming.

- Avotri. J. and V. Walters. 1999. "Just Look at Our Work and See if You Have Any Freedom on Earth: Ghanaian Women's Accounts of Their Work and Their Health", Social Science & Medicine, 48(12), 1123–1133.
- Banerjee, Abhijit, and Sendhil Mullainathan. 2010. The Shape of Temptation: Implications for the Economic Lives of the Poor, *NBER Working Paper* No. 15973.

Banerjee, Abhijit, Esther Duflo, and Nancy Qian. 2012. "On the Road: Access to Transportation Infrastructure and Economic Growth in China," NBER Working Paper No. 17897.

- Baum-Snow, Nathaniel, Loren Brandt, J. Vernon Henderson, Matthew A. Turner, and Qinghua Zhang. 2011. "Roads, Railways and Decentralization of Chinese Cities International", Growth Center Working Paper March, London, UK: London School of Economics.
- Behrman, Jere R, Susan W Parker, and Petra E Todd. 2008. "Medium-Term Impacts of the Oportunidades Conditional Cash Transfer Program on Rural Youth in Mexico," in *Poverty, Inequality, and Policy in Latin America*, 219–270.

Bimla, G.S., Dilbaghi, M., & Raina, K. 2003. "Rural Women Carry the Load of Fetching Water", Indian Journal of Social Work, 641, 65-75.

Borgatti, S. P., Everett, M. G., & Johnson, J. C. 2013. Analyzing Social Networks, Thousand Oaks, C.A.: Sage.

- Bucheli, J. R., Alok K. Bohara, and Kira Villa. 2016. "The Impact of a Rural Road Development Project on Multidimensional Poverty in Nepal", Paper presented at the Agricultural & Applied Economics Association Annual Meeting, Boston, Massachusetts, July 31–August 2.
- Buor, D. 2004. "Water Needs and Women's Health in the Kumasi Metropolitan Area, Ghana", Health and Place, 10(1), 85–103.
- Cascio, EU, and A Narayan. 2015. "Who Needs a Fracking Education? The Educational Response to Low-Skill Biased Technological Change," NBER Working Papers No. 21359.

Cervero, Ewing. 2010. "Travel and the Built Environment", Journal of the American Planning Association 76 (3), 265–294.

Cunha Flavio and Heckman James J. 2010. "Investing in Our Young People," NBER Working Papers No. 16201.

Del Carpio, Ximena, and Mywish Maredia. 2011. "Impact Evaluations in Agriculture. An Assessment of the Evidence." Independent Evaluation Group World Bank, Washington DC.

Dora C and Phillips M. 2000. Transport, Environment, and Health. World Health Organization. WHO Regional Publications, European Series, No. 89.

Douglas M, Thomson H, Jepson R, Hurley F, Higgins M, Muirie J, Gorman D. 2017. Health Impact Assessment of Transport Initiatives: A Guide. NHS Scotland Edinburgh.

Duflo, Esther. 2012. "Human Values and the Design of the Fight Against Poverty", The Tanner Lectures on Human Values, Harvard University.

Duflo, Esther, and Abhijit Banrjee. 2012. Poor Economics: A Radical Rethinking of the Way to Fight Global Poverty, New York, N.Y.: Public Affairs.

Duflo, Esther, Sebastian Galiani, and Mushfiq Mobarak. 2012. Improving Access to Urban Services for the Poor: Open Issues and a Framework for a Future Research Agenda, Cambridge: Poverty Action Lab.

- Duflo, Esther, Pascaline Dupas, and Michael Kremer. 2015. "Education, HIV, and Early Fertility: Experimental Evidence from Kenya", American Economic Review 105(9), 2757–2797.
- Egan, Matt, Mark Petticrew, David Ogilvie, and Val Hamilton, DipLib. 2009. "New Roads and Human Health: A Systematic Review", American Journal of Public Health, 93(9), 1463–147.
- Elvik Rune. 2009. The Power Model of the Relationship between Speed and Road Safety. Update and New Analyses. Report 1034/2009, Institute of Transport Economics, Oslo, Norway.
- Estache, Antonio. 2010. "A Survey of Impact Evaluation of Infrastructure Projects, Programs and Policies", ECORE Discussion Paper N.6. CORE and ECARES.
- Faber, Benjamin. 2014. "Trade Integration, Market Size, and Industrialization: Evidence from China's National Trunk Highway System," *Review of Economic Studies*, 81(3), 1046–1070.

Farrington D, Welsh B. 2007. "Improved Street Lighting and Crime Prevention". Swedish Council for Crime Prevention, Information, and Publications.

- Feigenberg, Benjamin, Erica Field, and Rohini Pande. 2013. "The Economic Returns to Social Interactions: Experimental Evidence from Microfinance", Review of Economic Studies, 80(4), 1459-1483.
- Ghani, Ejaz, Arti Grover Goswami, and William R. Kerr. 2016. "Highway to Success: The Impact of the Golden Quadrilateral Project for the Location and Performance of Indian Manufacturing," *Economic Journal*, 126(591), 317–357.

Gayen, Avijit, and Joydeep Chandra. 2010. "Role of Trust in Evolution of Scientific Collaboration Networks", Physical Review, 82(2), 026112.

Ghana Statistical Service (GSS). 2015. Ghana Demographic and Health Survey 2014, October, Accra, Ghana.

Haas, Ralph, Guy Felio, Zoubir Lounis, and Lynne Cowe Falls. 2009. "Measurable Performance Indicators for Roads: Canadian and International Practice", in Annual Conference of Transportation Association of Canada, Best Practices in Urban Transportation Planning: Measuring Change, Vancouver, B.C. October, pp. 1-22.

Hedges B. 1983. Adaptations to Traffic Noise. London, England: Social and Community Planning Research.

Heath, Rachel, and A. Mushfiq Mobarak. 2015. "Manufacturing Growth and the Lives of Bangladeshi Women," Journal of Development Economics, 115(1), 1–15.

Inter-American Development Bank (DB). 2018. "What is Sustainable Infrastructure? A Framework to Guide Sustainability Across the Project Cycle", Inter-American Development Bank IDB-Invest, Technical Note N IDB-TN-1388.

Jensen H.N. 1995. Vedrørende den Sikkerhedsmæssige Effect of Etablering af Motorveje. Copenhagen, Denmark: Sektorplannafdelingen Vejdirektoratet; 1995

Jensen, Roberts. 2012. "Do Labor Market Opportunities Affect Young Women's Work and Family Decisions? Experimental Evidence from India," The Quarterly Journal of Economics, 1272), 753–792.

Jonsson E, Sörensen. 1973. "Adaptation to Community Noise: a Case Study, Journal of Sound Vibration, 26(3), 571–575.

Juma, Calestous. 2011. The New Harvest: Agricultural Innovation in Africa, New York, N.Y.: Oxford University Press.

Jhurree, Vikashkumar. 2005. "Technology Integration in Education in Developing Countries: Guidelines to Policy Makers", International Education Journal, 6(4), 467-483.

Kawachi, I. 1999. "Social Capital and Community Effects on Population and Individual health." Annals New York Academy of Science, N. 896, 120-130.

- Khandker, Shahidur, Zaid Bakht, and Gayatri Koolwal. 2009. "The Poverty Impact of Rural Roads: Evidence from Bangladesh", *Economic Development and Cultural Change*, 57(4), 685-722.
- Kingombe, Christian. 2010. An Enquiry into the Causes and Nature of the Transmission Mechanisms between Labour-Based Rural Roads, Sustainable Growth, and Agricultural Trade in Zambia's Eastern Province. Ph.D. Thesis. The University of London.

Kingombe, Christian. 2015. "Steps in the Impact Evaluation of AfDB – Finance Rural Road Projects", eVALUatiOn Matters.

Kircher, S. 2007. "Hell on Earth – Systematic Rape in Eastern Congo", Journal of Humanitarian Assistance, 8.

- Miguel, Edward, and Michael Kremer. 2004. "Worms: Identifying Impacts on Education and Health in the Presence of Treatment Externalities," Econometrica, 72(1), 159–217.
- Morrissey C and B. Hedges. 1991. Adaptation to Increases in Traffic Exposure. TRRL Report 245, Crowthorne, England: Transport and Road Research Laboratory.
- Morten, Melanie, and Jaqueline Oliveira. 2018. "The Effects of Roads on Trade and Migration: Evidence from a Planned Capital City." NBER Working Paper No. 22158.
- Mu, Ren & van de Walle, Dominique. 2007. "Rural roads and poor area development in Vietnam," *Policy Research Working Paper Series N. 4340*, The World Bank.
- Nabors D, Schneider R, Leven D, Lieberman K, Mitchell C. 2008. *Pedestrian Safety Guide for Transit Agencies*. U.S Department of Transportation, Federal Highway Administration. FHWA-SA-07-017.
- Olken, Benjamin. 2007. "Monitoring Corruption: Evidence from a Field Experiment in Indonesia," Journal of Political Economy, 115(2).

Oxford Poverty and Human Development Initiative (OPHI). 2017. OPHI Country Briefing 2017: Ghana, University of Oxford.

Oster, Emily, and Bryce Steinberg. 2013. "Do IT Service Centers Promote Enrollment? Evidence from India," Journal of Development Economics, 103(1), 1–36.

- Partnership for Prevention (PFP). 2011. Transportation and Health: Policy Intervention for Safer, Healthier People and Communities. Washington, DC: Partnership for Prevention; 2011.
- Retting, R. A., S. A. Ferguson, and A. T. McCartt. 2003. "A Review of Evidence-Based Traffic Engineering Measures Designed to Reduce Pedestrian-Motor Vehicle Crashes", American Journal of Public Health 93(9), 1456-1463.

- Richards, D.C. 2010. *Relationship between Speed and Risk of Fatal Injury: Pedestrians and Car Occupants.* Transportation Research Laboratory. Road Safety Web Publication No. 16. Department for Transport: London.
- Rodrik, Dani. 2000. "Institutions for High-Quality Growth: What They Are and How to Acquire Them", Studies in Contemporary International Development, 35(3), 3-31.

Scott, John. 2008. Social Network Analysis: a Handbook. 2nd ed. CA: Sage. 2nd Edition.

- Shah, A. 2002. "Women, Water, Irrigation: Respecting Women's Priorities", Economic and Political Weekly, 37, 4413-4420.
- Shah, Manisha, and Bryce Millett Steinberg. 2015. "Workfare and Human Capital Investment: Evidence from India," NBER Working Papers Series No. 21543.
- Sorenson, Susan B., Christiaan Morssink, and Paola Abril Campos. 2011. "Safe Access to Safe Water in Low-Income Countries: Water Fetching in Current Times", Departmental Papers SPP, School of Social Policy and Practice, University of Pennsylvania.
- Steinmann, P., Keiser, J., Bos, R., Tanner, M., & Utzinger, J. 2006. "Schistosomiasis and Water Resources Development: Systematic Review, Meta-analysis, and Estimates of People at Risk. The Lancet Infectious Diseases, 63, 411-425.
- Storeygard, Adam. 2016. "Farther on Down the Road: Transport Costs, Trade and Urban Growth in Sub-Saharan Africa," *The Review of Economic Studies*, 83(3), 1263–1295.
- Taylor M.C., Lynam D.A., and Baruya A. 2000. "The Effects of Drivers' Speed on the Frequency of Road Accidents". *Transport Research Laboratory Report 421*. Crowthorne, UK: Transport Research Laboratory.
- United Nations. 2007. The Wye Group Handbook. Rural Households' Livelihood and Well-being, Statistics on Rural Development and Agricultural Household Income. N.Y.: New York.
- UNESCO Institute for Statistics (UNESCO). 2012. Ghana: Age Distribution and School Attendance of Girls Aged 9-13 Years, 20 December, Paris, France.
- UNICEF and WHO. 2008. Nations Children's Fund Joint Monitoring Programme for Water Supply and Sanitation, United Nations Children's Fund UNICEF) and the World Health Organization WHO), NY: New York.
- UNICEF and WHO. 2017. Progress on Household Drinking Water, Sanitation and Hygiene I 2000–2017, United Nations Children's Fund (UNICEF) and World Health Organization (WHO), NY: New York.
- Valdes-Vasquez, Rodolfo, and Leidy E. Klotz. 2013. "Social Sustainability Considerations during Planning and Design: Framework of Processes for Construction Projects Rodolfo", Journal of Construction Engineering and Management, pp. 80–89.
- Valente, Thomas W. 2010. Social Networks and Health: Models, Methods, and Application. Oxford: Oxford University Press
- van de Walle, Dominique. 2009. "Impact Evaluation of Rural Road Projects", Journal of Development Effectiveness, 1(1), 15-36.
- Vascocellos, Eduardo. 1997. "Rural Transport and Access to Education in Developing Countries: Policy Issues", Journal of Transport Geography, 5(2), 127–136
- Watson, M., and A. L. Dannenberg. 2008. "Investment in Safe Routes to School Projects: Public Health Benefits for the Larger Community." *Prevention of Chronic Diseases* 5(3).
- Weinstein, N.D. 1982. "Community Noise Problems: Evidence Against Adaptation", Journal of Environmental Psychology, 2(1), 87–97.
- World Health Organization (WHO). 2009. Global Status Report on Road Safety. Geneva: WHO.
- World Health Organization (WHO). 2009a. "Maternal Mortality Fact Sheet", Media Centre. November.
- World Health Organization (WHO). 2009b. "An Old Enemy Returns", Bulletin of the World Health Organization, 87, 85-86.
- Zwane, Alix Peterson, Jonathan Zinman, Eric Van Dusen, William Pariente, Clair Null, Edward Miguel, Michael Kremer, Dean S. Karlan, Richard Hornbeck, Xavier Giné, Esther Duflo, Florencia Devoto, Bruno Crepon, and Abhijit Banerjee. 2011 "Being Surveyed Can Change Later Behavior and Related Parameter Estimates", Proceedings of the National Academy of Science, 108(5), 1821-1826.

Endnotes

- 1 See, for instance: Baum-Snow et al. (2011); Egan et al. (2009); Kircher (2007); and Shah (2002).
- 2 In December 2018, two new regions (North East and Savannah) were carved out of the Northern Region after a referendum. Therefore, in this document we will use Savannah Region in lieu of Northern Region, which was the name of the region at the time of the construction of the Fufulso-Sawla Road.
- 3 AfDB (2015), p. 10.
- 4 For a complete description of all ancillary works see AfDB Aide Memoires of March 2014 through December 2016, as well as Ablin Consult (2015).
- 5 Ablin Consult (2015).
- 6 AfDB (2015), p.4.
- 7 The graphs present the density function in the Y axes and the mean of the variables plus values of the mean+1-3 STD DEV. Indeed values are between 1 and 4 for all the variables, but the graph reports, as it is standard, values for the referred MEAN+STD DEV (in reality it is 2.698*STD DEV, not 3).
- 8 Details of the elements included in the Index of quality water are included in Annex 5 of the Technical Annexes.
- 9 Corresponding to around US\$68 per month, at the monthly exchange rate as at 30 September 2019.







About this evaluation

This summary report presents the findings, conclusions, lessons, and recommendations of the impact evaluation of the AfDB's support for a road transport intervention in Ghana - the Fufulso-Sawla Road Project. The project, worth UA 110.58 million, was approved in 2010 and completed in 2015. It aimed at enhancing accessibility along the Fufulso-Sawla Road and improving livelihoods in the project area of influence. The project stands out as a flagship in terms of its inclusive and integrated design, which provides a holistic response to the socio-economic needs of the beneficiary districts.

The objectives of this impact evaluation were: (i) to estimate the impacts of AfDB supported integrated infrastructure projects on key intermediate and long-term outcomes; (ii) to identify explanatory factors that affect the development outcomes of the project; and (iii) to generate lessons and provide recommendations for improving the impacts of ongoing and future integrated infrastructure interventions.

As a result of the evaluation, IDEV drew lessons on integrating community development interventions into road transport infrastructure, the key factors for the sustainability of such projects, and the benefits for women and girls. It made the following recommendations: i) enhance the Bank's integrated approach to its road investments to foster development impact in terms of poverty reduction; ii) improve the quality of road projects' design and results-focus; and iii) strengthen the human and institutional capacity to sustain development gains.





Independent Development Evaluation African Development Bank

African Development Bank Group Avenue Joseph Anoma, 01 BP 1387, Abidjan 01, Côte d'Ivoire Phone: +225 20 26 28 41 E-mail: idevhelodesk@afdb.org



40 years of evaluation at the AfDE 40 ans d'évaluation à la BAD



idev.afdb.org